Existence of Equilibria in Games with Arbitrary Strategy Spaces and Payoffs: A Full Characterization

Guoqiang Tian* Department of Economics Texas A&M University College Station, Texas 77843 USA

November, 2008/Revised: May, 2009

Abstract

This paper provides a complete solution to the question of the existence of equilibria in games with general strategy spaces that may be discrete, continuum or non-convex and payoff functions that may be discontinuous or do not have any form of quasi-concavity. We establish a single condition, called *recursive diagonal transfer continuity*, which is both necessary and sufficient for the existence of pure strategy Nash equilibrium in games with arbitrary compact strategy spaces and payoffs. As such, our result strictly generalizes all the existing results on the existence of pure strategy Nash equilibrium. Moreover, recursive diagonal transfer continuity also permits full characterization of symmetric, mixed strategy, and Bayesian Nash equilibria in games with general strategy spaces and payoffs. The approach and main result developed in the paper can also allow us to ascertain the existence of equilibria in important classes of economic games. As an illustration, we show how they can be employed to fully characterize the existence of competitive equilibrium for economies with excess demand functions. The method of proof adopted to obtain our main result is also new and elementary — a non-fixed-point-theorem approach.

Keywords: existence of pure, symmetric, mixed strategy, and Bayesian Nash equilibrium; recursive diagonal transfer continuity; discontinuity; non-quasiconcavity; nonconvexity.

^{*}This is a research project working on for a long time. I thank Paulo Barelli, Xiaoyong Cao, Eric Maskin, David Rahman, Mingjun Xiao, and seminar participants at Texas A&M University for comments and suggestions. Financial support from the National Natural Science Foundation of China (NSFC-70773073) and the Program to Enhance Scholarly and Creative Activities at Texas A&M University as well as from Cheung Kong Scholars Program at the Ministry of Education of China is gratefully acknowledged. E-mail address: gtian@tamu.edu

1 Introduction

The notion of Nash equilibrium introduced by Nash (1950, 1951) is probably one of the most important solution concepts in game theory that has wide applications in almost all areas in economics. The early theorems of Nash (1950, 1951), Debreu (1952), and Fan (1953) reveal that games possess a pure strategy Nash equilibrium if (1) the strategy spaces are nonempty, convex and compact, and (2) players have continuous and quasiconcave payoff functions. The theorems say nothing about equilibrium in games with discontinuous and/or non-quasiconcave payoffs. Glicksberg (1952) shows that games with compact Hausdorff strategy spaces and continuous payoffs possess mixed strategy Nash equilibria.

Many economic models such as the classic price competition models of Bertrand (1883), Hotelling (1929), and auction models such as the one in Milgrom and Weber (1982), patent race models of Fudenberg et al. (1983), etc., however, frequently exhibit discontinuities or nonquasiconcavity in payoffs. Consequently, the standard theorems cannot be applied to establishing the existence of a pure or mixed strategy Nash equilibrium. While in many of these games equilibria can be constructed, there are other models in which this is not the case. Indeed, even for multi-dimensional auction models, the existence of an equilibrium in this type of games has been at issue since payoffs are not continuous, see Jackson (2005). Also, many economic models do not have convex strategy spaces, and payoff functions under consideration may not be quasiconcave or even do not have any form of quasi-concavity.

Accordingly, economists have been struggling to seek weaker conditions that can still guarantee the existence of an equilibrium. So far, two approaches have been adopted to weaken the continuity and/or quasiconcavity. The first approach is to relax the quasi-concavity of payoffs and convexity of strategy spaces. In a homogeneous product setting McManus (1964) and Roberts and Sonnenschein (1976) show the existence of a symmetric Cournot equilibrium allowing for a general downward sloping demand when there are n identical firms with convex costs. Nishimura and Friedman (1981) assume, in addition to some conditions on best reply correspondences, that payoff functions are continuous. Topkis (1979), Vives (1990), and Milgrom and Roberts (1990) establish the existence in games where payoffs are upper semi-continuous and satisfy certain monotonicity properties.

The central idea of this approach is based upon lattice-theoretical concepts, and at its heart lies Tarski's (1955) fixed point theorem. An advantage of Vives' approach is that it does not require the convexity of strategy sets. Payoffs need not be quasiconcave in all of the papers mentioned above, and additionally payoffs need not be continuous in some of them. The key property is that best-replies are increasing in the opponents' strategies, which guarantees the existence of pure strategy Nash equilibrium. However, the lattice-theoretic approach requires payoff functions must be upper semi-continuous in one's own strategy in order to guarantee the existence of best replies. This assumption fails to hold in virtually all auctions, as well as in the classic games of Bertrand and Hotelling.

The second approach is the topological approach proposed to weaken the continuity of payoff functions. Dasgupta and Maskin (1986) are the first to establish an existence theorem valid for games with discontinuous payoff functions. Their results reveal that such games possess a pure strategy equilibrium, provided (1) the strategy spaces are nonempty, convex and compact, and (2) players have payoff functions that are quasiconcave, upper semi-continuous, and graph continuous. They also investigate the existence of mixed strategy Nash equilibrium in games with discontinuous payoffs.

Simon (1987) obtains the existence of mixed strategy Nash equilibria in discontinuous games through introducing the notion of reciprocal upper semi-continuity (under the name of "complementary discontinuities") and thus strictly generalizes the result of Dasgupta and Maskin (1986). Reciprocal upper semi-continuity requires that some player's payoff jump up whenever some other player's payoff jumps down, which generalizes the condition that the sum of the players' payoffs is upper semicontinuous.

Simon and Zame (1990) establish the existence of a Nash equilibrium in mixed strategies with an endogenous sharing rule. While in some settings involving discontinuities this approach is remarkably helpful, in others it is less so. In an auction design environment where discontinuities are sometimes deliberately introduced, the participants must be presented with a game that fully describes the strategies and payoffs, since one cannot leave some of the payoffs unspecified or somehow endogenously determined. In addition, this method is only useful in establishing the existence of a mixed, as opposed to pure, strategy equilibrium.

Baye, Tian, and Zhou (1993) investigate the existence of pure strategy Nash equilibrium and dominant-strategy equilibrium by weakening both continuity and quasi-concavity of payoffs. It is shown that diagonal transfer quasi-concavity is necessary, and further, under diagonal transfer continuity and compactness, sufficient for the existence of pure strategy Nash equilibrium. Both diagonal transfer quasi-concavity and diagonal transfer continuity are very weak notions of quasi-concavity and continuity, which adopt a basic idea of transferring a set of strategy profile(s) to another set of strategy profile(s).

Reny (1999) establishes the existence of Nash equilibrium in compact and quasiconcave games that are better-reply secure, which is also a weak notion of continuity. Reny (1999) shows that better-reply security can be imposed separately as reciprocal upper semi-continuity introduced by Simon (1987) and payoff security. Bagh and Jofre (2006) further weaken reciprocal upper semicontinuity to weak reciprocal upper semi-continuity and show that it, together with payoff security, implies better-reply security. Both better-reply security and payoff security use a similar idea of transferring a (non-equilibrium) strategy to a securing strategy, and therefore they are also in the form of transfer continuity.

Nessah and Tian (2008) introduce an even weaker form of continuity, called *weak transfer quasi-continuity*, which is weaker than diagonal transfer continuity in Baye, Tian, and Zhou (1993) and better-reply security in Reny (1999), and holds in a large class of discontinuous games. They show that weak transfer quasi-continuity, together with the compactness of strategy spaces and quasiconcavity or (strong/weak) diagonal transfer quasiconcavity of payoffs, permits the existence of pure strategy Nash equilibrium so that it strictly generalizes the results of Baye, Tian, and Zhou (1993) and Reny (1999). They provide some sufficient conditions for weak transfer quasi-continuity by introducing notions of weak transfer continuity, weak transfer upper continuity and weak transfer lower continuity. These conditions are satisfied in many economic games and are often quite simple to check. They also study the existence of mixed strategy Nash equilibria in discontinuous and nonconvex games.

Recently, Barelli and Soza (2009) further significantly weaken the continuity and quasiconcavity conditions. They generalize most existing results and establish the existence of pure strategy Nash equilibria in discontinuous quasiconcave games and qualitative convex games.

However, all the existing results only give sufficient conditions for the existence of equilibrium, and no full characterization has been given yet in the literature.¹ The existing results use two separated conditions: continuity and quasi-concavity/monotonicity. Neither single unified condition nor full characterization approach has been given. A question is then whether or not there exists a single unified condition that can be used to prove the existence of (pure/mixed) Nash equilibrium in games with arbitrary strategy spaces and payoff functions, and if so, what the weakest condition is. This paper provides a complete answer to these questions by giving a necessary and sufficient condition for the existence of equilibrium in games with arbitrary strategy spaces and payoffs.

This paper fully characterizes the existence of pure strategy Nash equilibrium in games with general topological strategy spaces that may be discrete, continuum or non-convex and payoff functions that may be discontinuous or do not have any form of quasi-concavity. It is shown that the condition, *recursive (or called sequential) diagonal transfer continuity* introduced in the pa-

¹In the mechanism design literature, a lot of studies on full characterizations of Nash implementation of a social choice correspondence have been given such as those in Maskin (1999), Moore and Repullo (1990), Dutta and Sen (1991), etc. Also, Rahman (2008) recently provides a full characterization of correlated equilibrium.

per, is necessary and sufficient for the existence of pure strategy Nash equilibrium in games with arbitrary compact strategy spaces and payoff functions. As such, it strictly generalizes all the existing theorems on the existence of pure strategy Nash equilibrium. Recursive diagonal transfer continuity defined on respective spaces also permits full characterization of symmetric pure strategy, mixed strategy Nash, and Bayesian Nash equilibria in games with general strategy spaces and payoffs. Our full characterization result provides not only a way of understanding equilibrium, but also a way of checking the existence/nonexistence of pure strategy Nash equilibrium in games with discontinuous or nonconcave payoffs. We use quite a few known examples to illustrate the usefulness of our main theorem, especially its usefulness in checking the nonexistence of equilibrium. The approach and main result developed in the paper can also allow us to ascertain the existence of equilibria in important classes of economic games. As an application, we show how they can be employed to fully characterize the existence of competitive equilibrium with excess demand functions.

The logic of recursive diagonal transfer continuity, which generalizes the notion of "diagonal transfer continuity" to allow for recursive dominance (sequential security), can be roughly described as follows. Whenever a profile of strategies is not an equilibrium, there is a strategy profile that will be transferred to any finite set of "sequential securing strategies", each of which upsets deviation strategy profiles uniformly and locally. This means whenever a strategy profile x is not an equilibrium, there is a deviation strategy profile y and an open set of candidate strategy profiles containing x, all of which are dominated by any recursive deviation (sequential securing) strategy profiles that directly or indirectly dominates y.

The relation of the recursive diagonal transfer continuity and diagonal transfer continuity is somewhat similar to that of the weak axiom of revealed preference (WARP) and strong axiom of revealed preference (SARP) in the revealed preference theory on the rational behavior of individual decision making. Directly revealing a preference by WARP is not enough to fully reveal individuals' preferences, and then one may use SARP – recursive sequences of indirect revealed preferences (transitive closure) to fully reveal the rational behavior. Similarly, diagonal transfer continuity or better-reply security alone is not enough to guarantee the existence of Nash equilibrium -a description of rational behavior of individuals' strategic decision making, one then may need to use recursive diagonal transfer continuity to fully characterize the existence of equilibrium.

The method of proof employed to obtain our main result is new. While there are different ways of establishing the existence of Nash equilibria, all the existing proofs use the fixed-point-theorem related approaches. As such, previous techniques fail for two reasons. First, all the approaches in the literature either use two types of conditions: continuity and quasi-convexity, or only provide

sufficient conditions to guarantee the existence of a pure strategy Nash equilibrium. The presence of discontinuity and non-quasi-concavity in payoffs may preclude the existence of best replies so that best reply correspondences need not be nonempty-valued or convex-valued. Consequently, both lattice-theoretical techniques and topological techniques (standard application of Kakutani's fixed point theorem) to best reply correspondences fail. Second, to weaken the continuity, and thus, to obtain the existence of an equilibrium, the existing approaches have only considered a direct deviation from a non-equilibrium strategy. Moreover, a remarkable advantage of our proof is that it is simple and elementary without using advanced math.

The remainder of the paper is organized as follows. Section 2 describes the notation, and provides a number of preliminary definitions used in our study of noncooperative games, including the definition of the aggregate function that underlies our analysis of noncooperative games. Section 3 introduces the new condition, recursive diagonal transfer continuity, which is used in our full characterization of pure strategy Nash equilibrium. We prove our main result that recursive diagonal transfer continuity is a necessary and sufficient condition for the existence of pure strategy Nash equilibrium for arbitrary strategy spaces and payoffs. We also provide sufficient conditions for recursive diagonal transfer continuity to be true. Section 4 extends the full characterization result to symmetric pure strategy Nash equilibrium. Section 5 fully characterizes the existence of mixed strategy Nash equilibrium in games with arbitrary strategy spaces and payoffs. Section 6 shows recursive diagonal transfer continuity is also a necessary and sufficient condition for the existence of Bayesian Nash equilibrium in games. Section 7 shows how our main result can be employed to fully characterize the existence of competitive equilibrium for economies with excess demand functions. Finally, concluding remarks are offered in Section 8.

2 Preliminaries

Consider the following noncooperative game in the normal form:

$$G = (X_i, u_i)_{i \in I} \tag{1}$$

where $I = \{1, ..., n\}$ is the finite set of players,² X_i is player *i*'s strategy space which is a nonempty subset of a topological space E_i , and $u_i : X \longrightarrow \mathbb{R}$ is the payoff function of player *i*. Denote by $X = \prod_{i \in I} X_i$ the Cartesian product of the sets of strategy profiles of the game. For each player $i \in I$, denote by -i all other players rather than player *i*. Also denote by $X_{-i} = \prod_{j \neq i} X_j$ the Cartesian product of the sets of strategies of players -i.

²All the results in the paper hold for a countable infinity of players.

A strategy profile $x^* \in X$ is a *pure strategy Nash equilibrium* of a game G if,

$$u_i(y_i, x_{-i}^*) \le u_i(x^*) \; \forall i \in I, \; \forall y_i \in X_i.$$

A game $G = (X_i, u_i)_{i \in I}$ is compact, convex, bounded, and upper (lower) semi-continuous if, for all $i \in I$, X_i is compact, convex, and u_i is bounded and upper (lower) semi-continuous on X, respectively. A game $G = (X_i, u_i)_{i \in I}$ is quasiconcave if, for every $i \in I$, X_i is convex and the function u_i is quasiconcave in x_i .

To fully characterize the existence of equilibria, our strategy is to consider a mapping of individual payoffs into an aggregator function, and then provide a condition on the aggregator function that guarantees the existence of pure strategy Nash equilibrium. This kind of approach is pioneered by Nikaido and Isoda (1955), and is also used by Baye, Tian, and Zhou (1993). Dasgupta and Maskin (1986) also use a similar approach to prove the existence of mixed strategy Nash equilibrium in games with discontinuous payoff functions.

Our full characterization of pure strategy Nash equilibrium is based on the aggregator function, $U: X \times X \rightarrow \Re$ defined by³

$$U(y,x) = \sum_{i=1}^{n} u_i(y_i, x_{-i}), \quad \forall (x,y) \in X \times X,$$
(2)

which refers to the aggregate payoff across individuals where for every player *i* assuming she or he deviates to y_i given that all other players follow the strategy profile *x*. We may call U(y, x)the virtual deviation aggregate payoff, which may not be realizable. Immediately, we have the following observation.

LEMMA 2.1 $x^* \in X$ is a pure strategy Nash equilibrium of a game G if and only if $U(y, x^*) \leq U(x^*, x^*)$ for all $y \in X$.

Proof: Suppose $U(y, x^*) \leq U(x^*, x^*)$ for all $y \in X$. Let $y = (y_i, x^*_{-i})$. We then have

$$u_i(y_i, x_{-i}^*) \le u_i(x^*) \quad \forall \ y_i \in X,$$

$$(3)$$

which means that x^* is a Nash equilibrium. The converse is obvious by summing up (3) for all players.

³When I is a countably infinite set, one may define U according to $U(y, x) = \sum_{i \in I} \frac{1}{2^i} u_i(y_i, x_{-i})$. This is a more general formulation.

3 Full Characterization of Pure Strategy Nash Equilibria

In this section we provide a complete solution to the question of the existence of pure strategy Nash equilibrium in games with arbitrary compact strategy spaces and payoffs by providing a necessary and sufficient condition for the existence of pure strategy Nash equilibrium.

3.1 Main Result

We begin with stating diagonal transfer continuity introduced by Baye, Tian, and Zhou (1993) since the notion of recursive diagonal transfer continuity is somewhat an extension of diagonal transfer continuity.

DEFINITION 3.1 A game $G = (X_i, u_i)_{i \in I}$ is diagonally transfer continuous if, whenever U(y, x) > U(x, x) for $x, y \in X$, there exists a deviation strategy profile $z \in X$ and a neighborhood $\mathcal{V}_x \subset X$ of x such that U(z, x') > U(x', x') for all $x' \in \mathcal{V}_x$.

Note that "U(y,x) > U(x,x) for $x, y \in X$ " means " $x \in X$ is not an equilibrium". We will use these terms interchangeably. We say that a strategy profile y upsets strategy profile x if U(y,x) > U(x,x). Also, since the deviation strategy profile z results in a strictly higher payoff in a neighborhood of x, we call such a deviation strategy profile z a securing strategy profile. The diagonal transfer continuity then simply means that, whenever x is not an equilibrium, there exists a securing strategy profile z that upsets all points in some neighborhood of x. For convenience of exposition, let $U(z, V_x) > U(V_x, V_x)$ denote U(z, x') > U(x', x') for all $x' \in V_x$, where V_x is a neighborhood of x.

DEFINITION 3.2 (Recursive Upsetting) A strategy profile $y^0 \in X$ is said to be *recursively* (or called sequentially) upset by $z \in X$ if there exists a finite set of deviation strategy profiles $\{y^1, y^2, \ldots, y^{m-1}, z\}$ such that $U(y^1, y^0) > U(y^0, y^0), U(y^2, y^1) > U(y^1, y^1), \ldots, U(z, y^{m-1}) > U(y^{m-1}, y^{m-1}).$

We say that a strategy profile $y^0 \in X$ is *m*-recursively upset by $z \in X$ if the number of such deviation strategy profiles is *m*. For convenience, we say y^0 is directly upset by *z* when m = 1, and indirectly upset by *z* when m > 1. Recursive upsetting says that a strategy profile y^0 can be directly or indirectly upset by a strategy profile *z* through sequential deviation strategy profiles $\{y^1, y^2, \ldots, y^{m-1}\}$ in a recursive way that y^0 is upset by y^1, y^1 is upset by y^2, \ldots , and y^{m-1} is upset by *z*. The assertion that y^0 is directly upset by *z* is a dominance relation that the virtual aggregate payoff with individuals' deviation exceeds the realizable total payoff for a given strategy profile y^0 . Then the recursive upsetting says this dominance holds for any finite sequence of strategy profiles originated from a given strategy point.

We are now ready to introduce the notion of recursive diagonal transfer continuity.

DEFINITION 3.3 (Recursive Diagonal Transfer Continuity) A game $G = (X_i, u_i)_{i \in I}$ is said to be *recursively (or called sequentially) diagonal transfer continuous* if, whenever U(y, x) > U(x, x) for $x, y \in X$, there exists a strategy profile $y^0 \in X$ (possibly $y^0 = x$) and a neighborhood \mathcal{V}_x of x such that $U(z, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x)$ for any z that recursively upsets y^0 .

Similarly, we can define *m*-recursive diagonal transfer continuity. A game $G = (X_i, u_i)_{i \in I}$ is *m*-recursively diagonal transfer continuous if the phrase "for any *z* that recursively upsets y^{0} " in the above definition is replaced by "for any *z* that *m*-recursively upsets y^{0} ". Thus, a game $G = (X_i, u_i)_{i \in I}$ is recursively diagonal transfer continuous if it is *m*-recursively diagonal transfer continuous on *X* for all $m = 1, 2 \dots$

REMARK 3.1 Under recursive diagonal transfer continuity, when $U(z, y^{m-1}) > U(y^{m-1}, y^{m-1}), U(y^{m-1}, y^{m-2}) > U(y^{m-2}, y^{m-2}), \dots, U(y^1, y^0) > U(y^0, y^0)$, we have not only $U(z, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x)$, but also $U(y^{m-1}, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x), \dots, U(y^1, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x)$ since it is also k-recursively diagonal transfer continuous for k = 1, 2..., m-1. That means all of the points in \mathcal{V}_x are upset by the sequence of securing strategy profiles $\{y^1, \dots, y^{m-1}, y^m\}$ that directly or indirectly upset y^0 .

Recursive diagonal transfer continuity means that whenever x is not an equilibrium, there is a strategy profile y^0 and an open set of candidate strategy profiles containing x, all of which are upset by all securing strategy profiles that directly or indirectly upset y^0 . This implies that, if equilibrium fails to exist, then there is a nonequilibrium strategy profile x such that for every $y^0 \in X$ and every neighborhood \mathcal{V}_x of x, some deviation strategy profiles in the neighborhood cannot be upset by a securing strategy profile z that directly or indirectly upsets y^0 .

Recursive diagonal transfer continuity refers to the fact that, when U(y, x) > U(x, x), y may be transferred to a sequence of securing strategy profiles $\{y^1, y^2, \dots, y^m\}$ in order for all points in a neighborhood of x to be upset by these securing strategy profiles. The usual notion of continuity would require that this dominance hold at y for all points in a neighborhood of x.

REMARK 3.2 When m = 0, there is no recursive upsetting. 0-recursive diagonal transfer continuity becomes the diagonal transfer continuity whenever there is a securing strategy profile y^0 . Also, recursive diagonal transfer continuity neither implies nor is implied by continuity for games with two or more players.⁴ This point becomes clear when one sees recursive diagonal transfer continuity is a necessary and sufficient condition for the existence of pure strategy Nash equilibrium for arbitrary strategy spaces and payoff functions while continuity of the aggregate payoff function is not a necessary nor sufficient condition for the existence of pure strategy Nash equilibrium.

Before proceeding to our main result, we describe the basic idea why recursive diagonal transfer continuity ensures the existence of pure strategy Nash equilibrium for a compact game. When a compact game fails to have pure strategy Nash equilibrium, every strategy profile x will be upset by another strategy profile y^0 . Then, by recursive diagonal transfer continuity, there is some open set of candidate profiles containing x, all of which will be upset by some securing strategy profile z that directly or indirectly upsets y^0 . Then there are finite strategy profiles $\{x^1, x^2, \ldots, x^n\}$ whose neighborhoods cover X. Thus, all of the points in a neighborhood, say \mathcal{V}_{x^1} , will be upset by a corresponding deviation profile z^1 , which means z^1 cannot be an element in \mathcal{V}_{x^1} . If it is in some other neighborhood, say, \mathcal{V}_{x^2} , then it can be shown that z^2 will upset all strategy profiles in the union of \mathcal{V}_{x^1} and \mathcal{V}_{x^2} so that z^2 is not in the union of \mathcal{V}_{x^1} and \mathcal{V}_{x^2} . Suppose $z^2 \in \mathcal{V}_{x^3}$. Then we can similarly show that z^3 is not in the union of \mathcal{V}_{x^1} , \mathcal{V}_{x^2} and \mathcal{V}_{x^3} . Repeating such arguments, we can show that $z^k \notin \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2} \cup \ldots, \cup \mathcal{V}_{x^k}$, i..e, z^k is not in the union of \mathcal{V}_{x^1} , $\mathcal{V}_{x^2}, \ldots, \mathcal{V}_{x^k}$ for $k = 1, 2, \ldots, n$. In other words, $z^k \notin \mathcal{V}_{x^j}$ for all $j \leq k$ and $k = 1, 2, \ldots, n$ so that the securing strategy profile z^n will not be in the strategy space X, which is impossible. Thus recursive diagonal transfer continuity guarantees the existence of a pure strategy Nash equilibrium.

Now we are ready to state our main result that strictly generalizes all the existing results on the existence of pure strategy Nash equilibrium as special cases.

THEOREM 3.1 Suppose $G = (X_i, u_i)_{i \in I}$ is compact. Then, the game G possesses a pure strategy Nash equilibrium if and only if it is recursively diagonal transfer continuous on X.

PROOF. Sufficiency (\Leftarrow). Suppose, by way of contradiction, that there is no pure strategy Nash equilibrium. Then, by recursive diagonal transfer continuity, for each $x \in X$, there exists y^0 and a neighborhood \mathcal{V}_x such that $U(z, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x)$ whenever $y^0 \in X$ is recursively upset by z, i.e., for any sequence of recursive securing strategy profiles $\{y^1, \ldots, y^{m-1}, y^m\}$ with $U(y^m, y^{m-1}) > U(y^{m-1}, y^{m-1}), U(y^{m-1}, y^{m-2}) > U(y^{m-2}, y^{m-2}), \ldots, U(y^1, y^0) > U(y^0, y^0)$ for $m \ge 1$, we have $U(z, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x)$. Since there is no equilibrium by the contrapositive hypothesis and

⁴In one-player games recursive diagonal transfer continuity is equivalent to the player's utility function possessing a maximum on a compact set, and consequently it implies transfer weak upper continuity introduced in Tian and Zhou (1995), which is weaker than continuity.

the game is recursively diagonal transfer continuous on X, such a sequence of recursive securing strategy profiles $\{y^1, \ldots, y^{m-1}, y^m\}$ exists for some $m \ge 1$.

Since X is compact and $X \subseteq \bigcup_{x \in X} \mathcal{V}_x$, there is a finite set $\{x^1, \ldots, x^L\}$ such that $X \subseteq \bigcup_{i=1}^L \mathcal{V}_{x^i}$. For each of such x^i , the corresponding initial deviation profile is denoted by y^{0i} so that $U(z^i, \mathcal{V}_{x^i}) > U(\mathcal{V}_{x^i}, \mathcal{V}_{x^i})$ whenever y^{0i} is recursively upset by z^i .

Since there is no equilibrium, for each of such y^{0i} , there exists z^i such that $U(z^i, y^{0i}) > U(y^{0i}, y^{0i})$, and then, by 1-recursive diagonal transfer continuity, we have $U(z^i, \mathcal{V}_{x^i}) > U(\mathcal{V}_{x^i}, \mathcal{V}_{x^i})$. Now consider the set of securing strategy profiles $\{z^1, \ldots, z^n\}$. Then, $z^i \notin \mathcal{V}_{x^i}$, otherwise, by $U(z^i, \mathcal{V}_{x^i}) > U(\mathcal{V}_{x^i}, \mathcal{V}_{x^i})$, we will have $U(z^i, z^i) > U(z^i, z^i)$, a contradiction. So we must have $z^1 \notin \mathcal{V}(x^1)$.

Without loss of generality, we suppose $z^1 \in \mathcal{V}_{x^2}$. Since $U(z^2, z^1) > U(z^1, z^1)$ by noting that $z^1 \in \mathcal{V}_{x^2}$ and $U(z^1, y^{01}) > U(y^{01}, y^{01})$, then, by 2-recursive diagonal transfer continuity, we have $U(z^2, \mathcal{V}_{x^1}) > U(\mathcal{V}_{x^1}, \mathcal{V}_{x^1})$. Also, $U(z^2, \mathcal{V}_{x^2}) > U(\mathcal{V}_{x^2}, \mathcal{V}_{x^2})$. Thus $U(z^2, \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2}) > U(\mathcal{V}_{x^1} \cup \mathcal{V}_{x^2}, \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2})$, and consequently $z^2 \notin \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2}$.

Again, without loss of generality, we suppose $z^2 \in \mathcal{V}_{x^3}$. Since $U(z^3, z^2) > U(z^2, z^2)$ by noting that $z^2 \in \mathcal{V}_{x^3}$, $U(z^2, z^1) > U(z^1, z^1)$, and $U(z^1, y^{01}) > U(y^{01}, y^{01})$, by 3-recursive diagonal transfer continuity, we have $U(z^3, \mathcal{V}_{x^1}) > U(\mathcal{V}_{x^1}, \mathcal{V}_{x^1})$. Also, since $U(z^3, z^2) > U(z^2, z^2)$ and $U(z^2, y^{02}) > U(y^{02}, y^{02})$, by 2-recursive diagonal transfer continuity, we have $U(z^3, \mathcal{V}_{x^2}) >$ $U(\mathcal{V}_{x^2}, \mathcal{V}_{x^2})$. Thus, we have $U(z^3, \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2} \cup \mathcal{V}_{x^3}) > U(\mathcal{V}_{x^1} \cup \mathcal{V}_{x^2} \cup \mathcal{V}_{x^3}, \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2} \cup \mathcal{V}_{x^3})$, and consequently $z^3 \notin \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2} \cup \mathcal{V}_{x^3}$.

Applying repeatedly the above arguments, we can show that $z^k \notin \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2} \cup \ldots, \cup \mathcal{V}_{x^k}$, i.e., z^k is not in the union of $\mathcal{V}_{x^1}, \mathcal{V}_{x^2}, \ldots, \mathcal{V}_{x^k}$ for $k = 1, 2, \ldots, L$. In particular, for k = L, we have $z^L \notin \mathcal{V}_{x^1} \cup \mathcal{V}_{x^2} \ldots \cup \mathcal{V}_{x^L}$ and so $z^L \notin X$, a contradiction.

Necessity (\Rightarrow). Suppose x^* is a pure strategy Nash equilibrium and U(y, x) > U(x, x) for $x, y \in X$. Let $y^0 = x^*$ and \mathcal{V}_x be a neighborhood of x. Since $U(y, x^*) \leq U(x^*, x^*)$ for all $y \in Y$, it is impossible to find any sequence of strategy profiles $\{y^1, y^2, \ldots, y^m\}$ such that $U(y^1, y^0) > U(y^0, y^0), U(y^2, y^1) > U(y^1, y^1), \ldots, U(y^m, y^{m-1}) > U(y^{m-1}, y^{m-1})$. Hence, the recursive diagonal transfer continuity holds trivially.

Theorem 3.1 provides a necessary and sufficient condition for a game to have a pure strategy Nash equilibrium, which can be used to show the existence/non-existence of pure strategy Nash equilibrium for these games. Five examples that illustrate the usefulness of the above result will be given in Subsection 3.3 below.

In general, the weaker the conditions in an existence theorem are, the harder it is to verify whether the conditions are satisfied in a particular game. Although these examples show the usefulness of Theorem 3.1, especially in proving the nonexistence of pure strategy Nash equilibrium, given the generality of the condition, it is not surprising that the condition is, in general, not easy to check from a practical point of view. Since the main purpose of the paper is to characterize the existence of equilibria in games with general strategy spaces and payoffs, the condition does suggest a way to interpret equilibrium existence. It mainly shows what is possible for a game to have a pure strategy Nash equilibrium. Nevertheless, adding specificity to the model, we may get sufficient conditions for recursive diagonal transfer continuity, and consequently provide new sufficient conditions for the existence of pure strategy Nash equilibrium.

DEFINITION 3.4 (Deviation Transitivity) $G = (X_i, u_i)_{i \in I}$ is said to be *deviational transitive* if $U(y^2, y^1) > U(y^1, y^1)$ and $U(y^1, y^0) > U(y^0, y^0)$ imply that $U(y^2, y^0) > U(y^0, y^0)$. That is, the upsetting dominance relation is transitive.

COROLLARY 3.1 Suppose $G = (X_i, u_i)_{i \in I}$ is compact and deviational transitive. Then, there exists a pure strategy Nash equilibrium point if and only if G is 1-recursively diagonal transfer continuous.

PROOF. We only need to show that, when G is deviational transitive, 1-recursive diagonal transfer continuity implies m-recursive diagonal transfer continuity for $m \ge 1$. Suppose x is not an equilibrium. Then, by 1-recursive diagonal transfer continuity, there exists a strategy profile $y^0 \in X$ and a neighborhood \mathcal{V}_x of x such that $U(z, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x)$ whenever $U(z, y^0) > U(y^0, y^0)$ for any $z \in X$.

Now, for any sequence of deviation profiles $\{y^1, \ldots, y^{m-1}, y^m\}$, if $U(y^m, y^{m-1}) > U(y^{m-1}, y^{m-1}), U(y^{m-1}, y^{m-2}) > U(y^{m-2}, y^{m-2}), \ldots, U(y^1, y^0) > U(y^0, y^0)$, we then have $U(y^m, y^0) > U(y^0, y^0)$ by deviation transitivity of U, and thus by 1-recursive diagonal transfer continuity, $U(y^m, \mathcal{V}_x) > U(\mathcal{V}_x, \mathcal{V}_x)$. Since m is arbitrary, G is recursively diagonal transfer continuous.

3.2 Discussion and Related Work

The recursive diagonal transfer continuity we propose is of transfer type. The basic transfer method is developed in Tian (1992, 1993), Tian and Zhou (1992, 1995), Zhou (1992), and Baye, Tian, and Zhou (1993) for studying the maximization of binary relations that may be nontotal or nontransitive and the existence of equilibrium in games that may have discontinuous or nonquasi-concave payoffs. They develop three types of transfers: transfer continuities, transfer convexities, and transfer transitivities. This kind of properties have provided milestones in the literature on

the maximization of binary relations and the existence of equilibrium in games with discontinuous and/or nonquasiconcave payoffs. Various notions of transfer continuities, transfer convexities and transfer transitivities provide complete solutions to the question of the existence of maximal elements for complete preorders and interval orders — cf. Tian (1993) and Tian and Zhou (1995). The recursive diagonal transfer continuity proposed in this paper extends the static transfer approach to a dynamic (sequential) one so that it enables us to provide a complete solution to the question of the existence of equilibrium in games with arbitrary compact strategy spaces and payoffs.

We now discuss how Theorem 3.1 yields the results of Baye, Tian, and Zhou (1993), Reny (1999), Nessah and Tian (2008), and Barelli and Soza (2009).

Baye, Tian, and Zhou (1993) study the existence of pure strategy Nash equilibria in games with discontinuous and nonquasiconcave payoffs by introducing the concepts of diagonal transfer continuity and diagonal transfer qausiconcavity.

DEFINITION 3.5 A game $G = (X_i, u_i)_{i \in I}$ is diagonally transfer quasiconcave in y if, for any finite subset $Y^m = \{y^1, ..., y^m\} \subset X$, there exists a corresponding finite subset $X^m = \{x^1, ..., x^m\} \subset X$ such that for any subset $\{x^{k^1}, x^{k^2}, ..., x^{k^s}\} \subset X^m$, $1 \le s \le m$, and any $x \in co\{x^{k^1}, x^{k^2}, ..., x^{k^s}\}$, we have $\min_{1 \le l \le s} U(y^{k^l}, x) \le U(x, x)$.

Theorem 1 in Baye, Tian, and Zhou (1993) shows that diagonal transfer quasi-concavity is necessary, and further, under diagonal transfer continuity and compactness, sufficient for the existence of pure strategy Nash equilibrium.

Reny (1999) studies the existence of pure strategy Nash equilibria in discontinuous games by introducing the concepts of payoff security and better-reply security.

Let $\Gamma = \{(x, u) \in X \times \mathbb{R}^n : u_i(x) = u_i, \forall i \in I\}$ be the graph of the game. The closure of Γ in $X \times \mathbb{R}^n$ is denoted by $\overline{\Gamma}$. The frontier of Γ , which is the set of points in $\overline{\Gamma}$ but not in Γ , is denoted by Fr Γ .

DEFINITION 3.6 A game $G = (X_i, u_i)_{i \in I}$ is better-reply secure if, whenever $(x^*, u^*) \in \overline{\Gamma}$, x^* is not an equilibrium implies that there is some player $i, \overline{x}_i \in X_i$, and an open neighborhood $\mathcal{V}_{x_{-i}}$ of x_{-i} such that $u_i(\overline{x}_i, y_{-i}) > u_i(x^*)$ for all $y_{-i} \in \mathcal{V}_{x_{-i}}$.

The notion of better-reply security also uses the same idea of transferring a non-equilibrium strategy to a securing strategy, and thus it actually falls in the form of transfer continuity.

Theorem 3.1 in Reny (1999) shows that a game $G = (X_i, u_i)_{i \in I}$ possesses a Nash equilibrium if it is compact, bounded, quasiconcave and better-reply secure. Reny (1999) and Bagh and Jofre (2006) provide sufficient conditions for a game to be better-reply secure. **DEFINITION 3.7** A game $G = (X_i, u_i)_{i \in I}$ is payoff secure if for every player $i, x \in X$, and $\epsilon > 0$, there exists $\overline{x}_i \in X_i$ and an open neighborhood $\mathcal{V}_{x_{-i}}$ of x_{-i} such that $u_i(\overline{x}_i, y_{-i}) \ge u_i(x) - \epsilon$ for all $y_{-i} \in \mathcal{V}_{x_{-i}}$.

DEFINITION 3.8 A game $G = (X_i, u_i)_{i \in I}$ is reciprocally upper semicontinuous if, whenever $(x, u) \in \overline{\Gamma}$ and $u_i(x) \leq u_i$ for every player $i, u_i(x) = u_i$ for every player i.

DEFINITION 3.9 A game $G = (X_i, u_i)_{i \in I}$ is weakly reciprocal upper semi-continuous if, for any $(x, u) \in Fr \Gamma$, there is a player *i* and $\hat{x}_i \in X_i$ such that $u_i(\hat{x}_i, x_{-i}) > u_i$.

Reny (1999) shows that a game $G = (X_i, u_i)_{i \in I}$ is better-reply secure if it is payoff secure and reciprocally upper semi-continuous. Bagh and Jofre (2006) further show that $G = (X_i, u_i)_{i \in I}$ is better-reply secure if it is payoff secure and weakly reciprocal upper semi-continuous.

To provide more sufficient conditions for recursive diagonal transfer continuity, as a complement of this paper, Nessah and Tian (2008) recently introduce a new notion of weak continuity, called *weak transfer quasi-continuity*, which is weaker than the most known weak notions of continuity, including diagonal transfer continuity in Baye, Tian, and Zhou (1993) and better-reply security in Reny (1999), and holds in a large class of discontinuous games.

DEFINITION 3.10 A game $G = (X_i, u_i)_{i \in I}$ is said to be *weakly transfer quasi-continuous* if, whenever $x \in X$ is not an equilibrium, there exists a strategy profile $y \in X$ and a neighborhood $\mathcal{V}(x)$ of x so that for every $x' \in \mathcal{V}(x)$, there exists a player i such that $u_i(y_i, x'_{-i}) > u_i(x')$.

They also provide some sufficient conditions, each of which implies weak transfer quasicontinuity: (1) transfer continuity, (2) weak transfer continuity, (3) weak transfer upper continuity and payoff security, and (4) upper semicontinuity and weak transfer lower continuity.

DEFINITION 3.11 A game $G = (X_i, u_i)_{i \in I}$ is said to be *transfer continuous* if for all player *i*, whenever $u_i(z_i, x_{-i}) > u_i(x)$ for $z_i \in X_i$ and $x \in X$, there is some neighborhood $\mathcal{V}(x)$ of *x* and $y_i \in X_i$ such that $u_i(y_i, x'_{-i}) > u_i(x')$ for all $x' \in \mathcal{V}(x)$.

DEFINITION 3.12 A game $G = (X_i, u_i)_{i \in I}$ is said to be *weakly transfer continuous* if, whenever $x \in X$ is not an equilibrium, there exists a player $i, y_i \in X_i$ and a neighborhood \mathcal{V}_x of x such that $u_i(y_i, x'_{-i}) > u_i(x')$ for all $x' \in \mathcal{V}_x$.

DEFINITION 3.13 A game $G = (X_i, u_i)_{i \in I}$ is said to be *weakly transfer upper continuous* if, whenever $x \in X$ is not an equilibrium, there exists a player $i, \hat{x}_i \in X_i$ and a neighborhood \mathcal{V}_x of x such that $u_i(\hat{x}_i, x_{-i}) > u_i(x')$ for all $x' \in \mathcal{V}_x$. **DEFINITION** 3.14 A game $G = (X_i, u_i)_{i \in I}$ is said to be *weakly transfer lower continuous* if, whenever x is not a Nash equilibrium, there exists a player $i, y_i \in X_i$, and a neighborhood $\mathcal{V}_{x_{-i}}$ of x_{-i} such that $u_i(y_i, x'_{-i}) > u_i(x)$ for all $x'_{-i} \in \mathcal{V}_{x_{-i}}$.

They also introduce some new notions of transfer type of quasiconcavity, called *strong diagonal transfer quasiconcavity* and *weak diagonal transfer quasiconcavity*, each of which is a necessary condition for the existence of pure strategy Nash equilibrium, and consequently a necessary condition for recursive diagonal transfer continuity.

DEFINITION 3.15 A game $G = (X_i, u_i)_{i \in I}$ is said to be *strongly diagonal transfer quasi*concave if for any finite subset $\{y^1, ..., y^m\} \subset X$, there exists a corresponding finite subset $\{x^1, ..., x^m\} \subset X$ such that for any subset $\{x^{k^1}, x^{k^2}, ..., x^{k^s}\} \subset X^m$, $1 \leq s \leq m$, and any $x \in co\{x^{k^1}, x^{k^2}, ..., x^{k^s}\}$, there exists $y^h \in \{y^{k^1}, ..., y^{k^s}\}$ satisfying $u_i(y_i^{k^h}, x_{-i}) \leq u_i(x)$ for all $i \in I$.

DEFINITION 3.16 A game $G = (X_i, u_i)_{i \in I}$ is said to be *weakly diagonal transfer quasiconcave* if for any finite subset $\{y^1, ..., y^m\} \subset X$, there exists a corresponding finite subset $\{x^1, ..., x^m\} \subset X$ such that for each $\tilde{x} = \sum_{i,j} \lambda_{i,j} x^j \in \operatorname{co}\{x^h, h = 1, ..., m\}$, we have

$$\min_{(i,j)\in J} [u_i(y_i^j, \tilde{x}_{-i}) - u_i(\tilde{x})] \le 0,$$

where $\lambda_{i,j} \ge 0$ with $\sum_{i,j} \lambda_{i,j} = 1$ and $J = \{(i, j : \lambda_{i,j} > 0\}.$

It may be remarked that a game G is: (1) weakly transfer quasi-continuous if it is weakly transfer continuous, diagonally transfer continuous or better reply secure; (2) weakly transfer upper continuous if it is upper semicontinuous; and (3) weakly diagonal transfer quasiconcave and diagonally transfer quasiconcave if it is strongly diagonal transfer quasiconcave.

Nessah and Tian (2008) show that a game $G = (X_i, u_i)_{i \in I}$ possesses a Nash equilibrium if it is convex, compact, weakly transfer quasi-continuous, and quasiconcave or strongly/weakly diagonal transfer quasiconcave.

As a source of reference, we summarize these results as corollaries of Theorem 3.1 that provide sufficient conditions for recursive diagonal transfer continuity.

COROLLARY 3.2 Suppose $G = (X_i, u_i)_{i \in I}$ is compact and convex. Then any one of the following conditions is sufficient for a game to be recursively diagonal transfer continuous, and consequently it possesses a pure strategy Nash equilibrium:

- 1. the game is quasiconcave, upper semicontinuous in x_i , and group continuous in x^5 [Dasgupta and Maskin (1986)];
- 2. the game is diagonally transfer quasiconcave and diagonally transfer continuous [Baye, Tian and Zhou (1993)];
- 3. the game is bounded, quasiconcave, and better-reply secure [Reny (1999)];
- 4. the game is bounded, quasiconcave, payoff secure, and reciprocally upper semicontinuous [Reny (1999)];
- 5. the game is bounded, quasiconcave, payoff secure, and weakly reciprocal upper semicontinuous [Bagh (2006)];
- 6. the game is bounded, quasiconcave, and weakly transfer quasi-continuous [Nessah and Tian (2008)];
- 7. the game is strongly diagonal transfer quasiconcave and weakly transfer quasi-continuous [Nessah and Tian (2008)];
- 8. the game is bounded, weakly diagonal transfer quasiconcave, and weakly transfer quasicontinuous [Nessah and Tian (2008)];
- 9. the game is bounded, weakly diagonal transfer quasiconcave, and diagonally transfer continuous [Nessah and Tian (2008)];
- 10. the game is bounded, weakly diagonal transfer quasiconcave, and better-reply secure [Nessah and Tian (2008)];
- 11. the game is bounded, weakly diagonal transfer quasiconcave, weakly transfer upper continuous, and payoff secure [Nessah and Tian (2008)];
- 12. the game is bounded, weakly diagonal transfer quasiconcave, weakly transfer lower continuous, and upper semicontinuous [Nessah and Tian (2008)].

In addition, the conditions imposed in Nash (1951), Debreu (1952), Nikaido and Isoda (1955), Nishimura and Friedman (1981), Dasgupta and Maskin (1986), Vives (1990), Carmona (2005), Morgan and Scalzo (2007), etc., which guarantee the existence of pure strategy Nash equilibria, can be regarded as sufficient conditions of recursive diagonal transfer continuity.

⁵A payoff function $u_i : X \to \mathbb{R}$ is graph-continuous if for all $x \in X$, there exists a function $F_i : A_{-i} \to A_i$ with $F_i(\bar{x}_{-i}) = \bar{x}_i$ such that $u_i(F_i(x_{-i}), x_{-i})$ is continuous at $x_{-i} = \bar{x}_{-i}$.

REMARK 3.3 From Baye, Tian, and Zhou (1993), one knows that a game is diagonally transfer quasiconcave if any one of the following conditions is satisfied: (1) each $u_i(x_i, x_{-i})$ is concave in x_i ; (2) U(x, y) is concave in x; (3) U(x, y) is quasiconcave in x; and (4) U(x, y) is diagonally quasiconcave in x. A game is diagonally transfer continuous if any one of the following conditions is satisfied: (1) each $u_i(x_i, x_{-i})$ is continuous; (2) each $u_i(x_i, x_{-i})$ is upper semicontinuous in x_i and continuous in x_{-i} ; (3) U(x, y) is continuous; and (4) $\phi(x, y) \equiv U(x, y) - U(y, y)$ is lower semicontinuous in y. As a result, any pair of conditions, one each from the sufficient conditions for diagonal transfer quasi-concavity and diagonal transfer continuity, are sufficient for the existence of a pure strategy Nash equilibrium for games with compact strategy spaces.

3.3 Examples

In this subsection we provide five examples to illustrate how our main theorem fully characterizes the existence or nonexistence of pure strategy Nash equilibrium in games that do not satisfy the conditions of existing theorems. Examples 1 and 2 are games with pure strategy equilibria that are accounted for by our Theorem 3.1, but which violate the conditions of existing theorems. Examples 3-5 are games that do not have an equilibrium because they fail to satisfy our recursive diagonal transfer continuity condition.

EXAMPLE 3.1 Consider a game with n = 2, $X_1 = X_2 = [0, 1]$, and the payoff functions are defined by

$$u_i(x_1, x_2) = \begin{cases} 1 & \text{if } (x_1, x_2) \in \mathbb{Q} \times \mathbb{Q} \\ 0 & \text{otherwise} \end{cases} \quad i = 1, 2,$$

where $\mathbb{Q} = \{x \in [0, 1] : x \text{ is a rational number}\}.$

Then the game is compact and convex, but not quasiconcave. It is not weakly transfer quasicontinuous either (so it is not diagonally transfer continuous, better-reply secure, or weakly transfer continuous either). To see this, consider any nonequilibrium x that consists of irrational numbers. Then, for any neighborhood \mathcal{V}_x of x, choosing $x' \in \mathcal{V}_x$ with $x'_1 \in \mathbb{Q}$ and $x'_2 \in \mathbb{Q}$, we have $u_1(y_1, x'_2) \leq u_1(x'_1, x'_2) = 1$ and $u_2(x'_1, y_2) \leq u_2(x'_1, x'_2) = 1$ for any $y \in X$. So the game is not weakly transfer quasi-continuous. Thus, there is no existing theorem that can be applied.

However, it is recursively diagonal transfer continuous on X. Indeed, suppose U(y,x) > U(x,x) for $x = (x_1, x_2) \in X$ and $y = (y_1, y_2) \in X$. Let y^0 be any vector with rational numbers and \mathcal{V}_x be a neighborhood of x. Since $U(y, y^0) \leq U(y^0, y^0)$ for all $y \in Y$, it is impossible to have $\{y^1, y^2, \ldots, y^m\}$ such that $U(y^m, y^{m-1}) > U(y^{m-1}, y^{m-1}), U(y^{m-1}, y^{m-2}) > U(y^{m-1}, y^{m-1}) > U(y^{m-1}, y^{m-1})$

 $U(y^{m-2}, y^{m-2}), \ldots, U(y^1, y^0) > U(y^0, y^0)$ for any of such strategy profiles. Hence, the recursive diagonal transfer continuity holds. Therefore, by Theorem 3.1, this game has a pure strategy Nash equilibrium. In fact, the set of pure strategy Nash equilibria consists of all rational numbers on [0, 1].

EXAMPLE 3.2 Consider the two-player game with the following payoff functions defined on $[0,1] \times [0,1]$ studied by Barelli and Soza (2009).

$$u_{i}(x_{i}, x_{-i}) = \begin{cases} 0 & \text{if } x_{i} \in (0, 1) \\ 1 & \text{if } x_{i} = 0 \text{ and } x_{-i} \in \mathbb{Q} \\ 1 & \text{if } x_{i} = 1 \text{ and } x_{-i} \notin \mathbb{Q} \\ 0 & \text{otherwise} \end{cases},$$

where $\mathbb{Q} = \{x \in [0, 1] : x \text{ is a rational number}\}.$

This game is convex, compact, bounded and quasiconcave, but it is not weakly transfer quasicontinuous, and consequently, it is not diagonally transfer continuous, better-reply secure, or weakly transfer continuous, either. Thus, there is no existing theorem that can be applied.

To see the game is not weakly transfer quasi-continuous, consider the nonequilibrium x = (1,1). We then cannot find any $y \in X$ and any neighborhood $\mathcal{V}_{(1,1)}$ of (1,1) such that for every $x' \in \mathcal{V}_x$, there is a player *i* with $u_i(y_i, x'_{-i}) > u_i(x')$. We show this by considering two cases.

Case 1. $y_2 \neq 0$. Then, for any neighborhood $\mathcal{V}_{(1,1)}$ of (1,1), choosing $x' \in \mathcal{V}_x$ with $x'_1 = 1$ and $x'_2 \notin \mathbb{Q}$, we have $u_1(y_1, x'_2) \leq u_1(x'_1, x'_2) = 1$ and $u_2(x'_1, y_2) = u_2(x'_1, x'_2) = 0$.

Case 2. $y_2 = 0$. When $y_1 \neq 0$, choosing $x' \in \mathcal{V}_x$ with $x'_2 = 1$ and $x'_1 \notin \mathbb{Q}$, we have $u_1(y_1, x'_2) = u_1(x'_1, x'_2) = 0$ and $u_2(x'_1, y_2) \leq u_2(x'_1, x'_2) = 1$. When $y_1 = 0$, choosing $x' \in \mathcal{V}_x$ with $x'_1 \notin \mathbb{Q}$ and $x'_2 \notin \mathbb{Q}$, we have $u_1(y_1, x'_2) = u_1(x'_1, x'_2) = 0$ and $u_2(x'_1, y_2) = u_2(x'_1, x'_2) = 0$.

Thus, the game is not weakly transfer quasi-continuous, and Theorems 3.1-3.3 of Nessah and Tian (2008) can not be applied.

However, it is recursively diagonal transfer quasi-continuous. Indeed, suppose U(y,x) > U(x,x) for $x = (x_1, x_2) \in X$ and $y = (y_1, y_2) \in X$. Let $y^0 = (0,0)$ and \mathcal{V}_x be a neighborhood of x. Since $U(y, y^0) \leq U(y^0, y^0)$ for all $y \in Y$, it is impossible to have $\{y^1, y^2, \ldots, y^m\}$ such that $U(y^m, y^{m-1}) > U(y^{m-1}, y^{m-1}), U(y^{m-1}, y^{m-2}) > U(y^{m-2}, y^{m-2}), \ldots, U(y^1, y^0) > U(y^0, y^0)$ for any of such strategy profiles. Hence, the recursive diagonal transfer continuity holds. Therefore, by Theorem 3.1, this game has a pure strategy Nash equilibrium.

Our full characterization result is especially useful to check the nonexistence of equilibrium of economic games. No such result is available in the literature.

EXAMPLE 3.3 (DASGUPTA AND MASKIN) Consider the following game studied by Dasgupta and Maskin (1986): n = 2, $X_1 = X_2 = [0, 1]$, and the payoff functions are defined by

$$u_i(x_1, x_2) = \begin{cases} 0 & \text{if } x_1 = x_2 = 1 \\ x_i & \text{otherwise} \end{cases}$$
 $i = 1, 2.$

The game is not recursively diagonal transfer continuous on X. To see this, for x = (1,1)and $y \in (0,1) \times (0,1)$, we have U(y,x) > U(x,x). We then cannot find any $y^0 \in X$ and neighborhood \mathcal{V}_x of x such that U(z,x') > U(x',x') for every deviation profile z that is upset directly or indirectly by y^0 for all $x' \in \mathcal{V}_x$. We show this by considering two cases.

Case 1. $y^0 \neq (1,1)$. Then, for any neighborhood $\mathcal{V}_{(1,1)}$ of (1,1), choosing strategy profiles $z \in X$ and $x' \in \mathcal{V}_{(1,1)}$ such that $y_1^0 + y_2^0 < z_1 + z_2 < x'_1 + x'_2$, we then have $U(z, y^0) > U(y^0, y^0)$ but U(z, x') < U(x', x').

Case 2. $y^0 = (1, 1)$. Then, for any neighborhood $\mathcal{V}_{(1,1)}$ of (1, 1), choosing strategy profiles $z \in X$ and $x' \in \mathcal{V}_{(1,1)}$ such that $0 < z_1 + z_2 < x'_1 + x'_2 < y_1^0 + y_2^0$, we then have $U(z, y^0) > U(y^0, y^0)$ but U(z, x') < U(x', x').

Thus, we cannot find any $y^0 \in X$ and any neighborhood \mathcal{V}_0 of (1, 1) such that U(z, x') > U(x', x') for every deviation profile z that is upset by y^0 for all $x' \in \mathcal{V}_x$. Hence, the game is not recursively diagonal transfer continuous on X, and therefore, by Theorem 3.1, there is no pure strategy Nash equilibrium on X.

EXAMPLE 3.4 (KARLIN) Consider games of "timing" or "silent duel", which have been studied by Karlin (1959), Owen (1968), Jones (1980), and Dasgupta and Maskin (1986). These are symmetric two-person zero-sum games on the unit square so that n = 2, $X_1 = X_2 = [0, 1]$, and U(x, x) = 0 for all $x \in X$. The version called the "silent duel" has player I's payoff function of the form:

$$u_1(x_1, x_2) = \begin{cases} x_1 - x_2 + x_1 x_2, & \text{if } x_1 < x_2 \\ 0, & \text{if } x_1 = x_2 \\ x_1 - x_2 - x_1 x_2, & \text{if } x_1 > x_2 \end{cases}$$

Consider $x = (x_1, x_2) = (1, 1)$. It can be verified that U(y, x) > U(x, x) implies that y must satisfy one of the following three sets of conditions: (1) $y_1 + y_2 > 1$, $y_1 < 1$, and $y_2 < 1$; (2) $y_1 = 1$ and $y_2 > 1/2$, and (3) $y_2 = 1$ and $y_1 > 1/2$. We then cannot find any $y^0 \in X$ and neighborhood \mathcal{V}_x of x such that U(z, x') > U(x', x') for every deviation profile z that is upset directly or indirectly by y^0 for all $x' \in \mathcal{V}_x$. To show this, four cases need to be considered.

Case 1. $y_1^0 < 1$ and $y_2^0 < 1$. Then, for any neighborhood $\mathcal{V}_{(1,1)}$ of (1,1), choose strategy profiles $z \in X$ and $x' \in \mathcal{V}_{(1,1)}$ such that $y_1^0 < z_1 < x'_1$ and $y_2^0 < z_2 < x'_2$. Since

 $u_1(y_1, y_2)$ and $u_2(y_1, y_2) = -u_1(y_1, y_2)$ are both increasing in y_1 and y_2 , respectively, we have $u_1(z_1, y_2^0) - u_1(y_1^0, y_2^0) > 0$ and $u_2(y_1^0, z_2) - u_2(y_1^0, y_2^0) = u_1(y_1^0, y_2^0) - u_1(y_1^0, z_2) > 0$, $u_1(z_1, x'_2) - u_1(x'_1, x'_2) < 0$ and $u_2(x'_1, z_2) - u_2(x'_1, x'_2) = u_1(x'_1, x'_2) - u_1(x'_1, z_2) < 0$. Thus, we have $U(z, y^0) > U(y^0, y^0)$ but U(z, x') < U(x', x').

Case 2. $y_1^0 = 1$ and $y_2^0 < 1$. Then, for any neighborhood $\mathcal{V}_{(1,1)}$ of (1,1), choose strategy profiles $z \in X$ and $x' \in \mathcal{V}_{(1,1)}$ such that $y_1^0 = z_1 = x'_1$ and $y_2^0 < z_2 < x'_2$. Then, by the monotonicity of $u_1(y_1, y_2)$ and $u_2(y_1, y_2) = -u_1(y_1, y_2)$, we have $U(z, y^0) > U(y^0, y^0)$ but U(z, x') < U(x', x').

Case 3. $y_1^0 < 1$ and $y_2^0 = 1$. Then, for any neighborhood $\mathcal{V}_{(1,1)}$ of (1,1), choose strategy profiles $z \in X$ and $x' \in \mathcal{V}_{(1,1)}$ such that $y_1^0 < z_1 < x'_1$ and $y_2^0 = z_2 = x'_2$. Then, by similar reasoning, we have $U(z, y^0) > U(y^0, y^0)$ but U(z, x') < U(x', x').

Case 4. $y_1^0 = 1$ and $y_2^0 = 1$. Then, for any neighborhood $\mathcal{V}_{(1,1)}$ of (1,1), choose strategy profiles $z \in X$ and $x' \in \mathcal{V}_{(1,1)}$ such that $1/2 < z_1 < x'_1$ and $1/2 = z_2 = x'_2$. We then have $u_1(z_1, y_2^0) - u_1(y_1^0, y_2^0) = 2z_1 - 1 > 0$ and $u_2(y_1^0, z_2) - u_2(y_1^0, y_2^0) = u_1(y_1^0, y_2^0) - u_1(y_1^0, z_2) = 2z_2 - 1 > 0$, $u_1(z_1, x'_2) - u_1(x'_1, x'_2) < 0$ and $u_2(x'_1, z_2) - u_2(x'_1, x'_2) = u_1(x'_1, x'_2) - u_1(x'_1, z_2) < 0$, and consequently, $U(z, y^0) > U(y^0, y^0)$ but U(z, x') < U(x', x').

Thus, we cannot find any $y^0 \in X$ and any neighborhood \mathcal{V}_0 of (1,1) such that U(z,x') > U(x',x') for every deviation profile z that is upset by y^0 for all $x' \in \mathcal{V}_x$. Hence, the game is not recursively diagonal transfer continuous on X, and therefore, by Theorem 3.1, there is no pure strategy Nash equilibrium on X.

The version called the "noisy duel" has player I's payoff function of the form:

$$u_1(x_1, x_2) = \begin{cases} 2x_1 - 1, & \text{if } x_1 < x_2 \\ 0, & \text{if } x_1 = x_2 \\ 1 - 2x_2, & \text{if } x_1 > x_2 \end{cases}$$

In this game, the payoff function $u_i(x_1, x_2)$ is neither diagonally transfer continuous nor quasiconcave in y_i for i = 1. Therefore, theorems in Baye, Tian, and Zhou (1993) and Reny (1999) are not applicable.

Suppose U(y,x) > U(x,x) for $x = (x_1,x_2) \in X$ and $y = (y_1,y_2) \in X$. Let $y^0 = (1/2,1/2)$ and \mathcal{V}_x be a neighborhood of x. Since $U(y,y^0) \leq U(y^0,y^0)$ for all $y \in Y$, it is impossible to have $\{y^1, y^2, \ldots, y^m\}$ such that $U(y^m, y^{m-1}) > U(y^{m-1}, y^{m-1})$, $U(y^{m-1}, y^{m-2}) > U(y^{m-2}, y^{m-2}), \ldots, U(y^1, y^0) > U(y^0, y^0)$ for any of such strategy profiles. Hence, the recursive diagonal transfer continuity holds. Therefore, by Theorem 3.1, this game has a pure strategy Nash equilibrium.

EXAMPLE 3.5 (VARIAN) Consider a two-person game with nonnegative price strategies p_1 and p_2 . Thus $Z_1 = Z_2 = [0, r]$ with r > 0. The payoffs $u_i(p_1, p_2)$ are given by the functions

$$u_i(p_1, p_2) = \begin{cases} p_i(I+\mu)k & \text{if } p_i < p_i \\ p_i(\frac{I}{2}+\mu)k & \text{if } p_1 = p_2 \\ p_i\mu k & \text{if } p_i > p_i \end{cases} \quad i = 1, 2.$$

This game has a number of interpretations. Varian (1980) interprets I to be the number of informed consumers, who will shop at the firm charging the lowest price, while 2μ is the number of uninformed consumers, who allocate themselves equally across the two firms. Thus each firm sells to μ uninformed consumers automatically, but gets the I informed consumers only if it succeeds in setting the lowest price.

It is well known that this game has no pure strategy Nash equilibrium (cf. Varian (1980); Baye, Kovenock, and de Vries (1992)), and in fact we can similarly verify that U is not recursively diagonal transfer continuous. Thus, by Theorem 3.1, the blame unambiguously lies squarely on the fact that the game is not recursively diagonal transfer continuous.

4 Full Characterization of Symmetric Pure Strategy Nash Equilibria

The techniques developed in the previous section can be used to fully characterize the existence of symmetric pure strategy Nash equilibrium. Throughout this section, we assume that the strategy spaces for all players are the same. As such, let $X_0 = X_1 = \ldots = X_n$. If in addition, $u_1(y, x, \ldots, x) = u_2(x, y, x, \ldots, x) = \ldots, u_n(x, \ldots, x, y)$ for all $x, y \in X$, we say that $G = (X_i, u_i)_{i \in I}$ is a quasi-symmetric game.

DEFINITION 4.1 A Nash equilibrium (x_1^*, \ldots, x_n^*) of a game G is said to be symmetric if $x_1^* = \ldots = x_n^*$.

For convenience, we denote, for each player *i*, and for all $x, y \in X_0, u_i(x, \ldots, y, \ldots, x)$ the function u_i evaluated at the strategy in which player *i* chooses *y* and all others choose *x*.

Define a quasi-symmetric function $\psi: X_0 \times X_0 \to \mathbb{R}$ by

$$\psi(y,x) = u_i(x,\dots,y,\dots,x). \tag{4}$$

Since G is quasi-symmetric, x^* is a symmetric pure strategy Nash equilibrium if and only if $\psi(y, x^*) \le \psi(x^*, x^*)$ for all $y \in X_i$.

We then have the following theorem.

THEOREM 4.1 Suppose a game $G = (X_i, u_i)_{i \in I}$ is quasi-symmetric and compact. Then it possesses a symmetric pure strategy Nash equilibrium if and only if $\psi(y, x)$ defined by (4) is recursively diagonal transfer continuous on X.

PROOF. The proof is the same as that of Theorem 3.1 provided U is replaced by ψ , thus it is omitted here.

Theorem 4.1 strictly generalizes all the existing results on the existence of symmetric pure strategy Nash equilibrium such as those in Reny (1999).

EXAMPLE 4.1 (BAGH AND JOFRE) The following two-person concession quasi-symmetric game on the unit square considered by Bagh and Jofre (2006) is a special case of a class of timing games on the unit square considered by Reny (1999). The payoffs are:

$$u_i(x_1, x_2) = \begin{cases} 10, & \text{if } x_i < x_{-i} \\ 1, & \text{if } x_i = x_{-i} < 0.5 \\ 0, & \text{if } x_i = x_{-i} \ge 0.5 \\ -10, & \text{if } x_i > x_{-i} \end{cases}$$

Note that the payoffs are not quasiconcave (nor are they quasiconcave along the diagonal of the unit square). We now show that ψ is recursively diagonal transfer continuous, and thus the game possesses a symmetric pure strategy equilibrium. Indeed, let $\psi(y, x) = u_i(y, x)$. Suppose $\psi(y, x) > \psi(x, x)$ for $x = (x_1, x_2) \in X$ and $y = (y_1, y_2) \in X$. Let $y^0 = (0, 0)$ and \mathcal{V}_x be a neighborhood of x. It is clear that $\psi(y, y^0) \leq \psi(y^0, y^0)$ for all $y \in Y$, and thus it is impossible to have $\{y^1, y^2, \ldots, y^m\}$ such that $\psi(y^m, y^{m-1}) > \psi(y^{m-1}, y^{m-1}), \psi(y^{m-1}, y^{m-2}) > \psi(y^{m-2}, y^{m-2}), \ldots, \psi(y^1, y^0) > \psi(y^0, y^0)$ for any of such strategy profiles. Hence, the recursive diagonal transfer continuity holds, and thus by Theorem 4.1, this game has a pure strategy Nash equilibrium.

EXAMPLE 4.2 (HENDRICKS AND WILSON) Consider the concession quasi-symmetric game between two players studied by Hendricks and Wilson (1983), Simon (1987), and Reny (1999). The players must choose a time $x_1, x_2 \in [0, 1]$ to quit the game. The player who quits last wins, although conditional on winning, quitting earlier is preferred. If both players quit at the same time, the unit prize is divided evenly between them. Then payoffs are:

$$u_i(x_1, x_2) = \begin{cases} -x_i, & \text{if } x_i < x_{-i} \\ 1/2 - x_i, & \text{if } x_i = x_{-i} \\ 1 - x_i, & \text{if } x_i > x_{-i} \end{cases}$$

Note that the payoffs are not quasiconcave (nor are they quasiconcave along the diagonal of the unit square) although U is diagonally transfer continuous by Proposition 2.(e) in Baye, Tian, and Zhou (1993). We now show that ψ is not recursively diagonal transfer continuous, and thus the game does not possess a symmetric pure strategy equilibrium. To see this, consider x = 0. It is clear that $\psi(y, x) = u_i(y, 0) > u_i(0, 0)$ implies that 0 < y < 1/2. We then cannot find any $y^0 \in X_0$ and neighborhood \mathcal{V}_x of x such that $\psi(z, x') > \psi(x', x')$ for every deviation profile z that is upset by y^0 for all $x' \in \mathcal{V}_x$. We show this by considering two cases.

Case 1. $y^0 \neq 0$. Then, for any neighborhood \mathcal{V}_0 of 0, choose a strategy profile $z \in [0, 1]$ and a strategy profile $x' \in \mathcal{V}_0$ such that $\max\{1/2 + \epsilon, y^0\} < z < 1/2 + y^0$ and $x' < \epsilon$, where $0 < \epsilon < \min\{1/2, y^0\}$. Then, by $z > y^0$ and $1 - z > 1/2 - y^0$, we have $\psi(z, y^0) > \psi(y^0, y^0)$. However, since z > x' and $1/2 + x' < 1/2 + \epsilon < z$, we have 1 - z < 1/2 - x', and consequently $\psi(z, x') < \psi(x', x')$.

Case 2. $y^0 = 0$. Note that $\psi(z, y^0) > \psi(y^0, y^0)$ if and only if 0 < z < 1/2. Then, for any neighborhood \mathcal{V}_0 of 0, choosing a positive number ϵ such that $(\epsilon/2, \epsilon) \subset \mathcal{V}_0$, $z = \epsilon/2$ and a strategy profile $x' \in \mathcal{V}_0$ such that $x' \in (\epsilon/2, \epsilon)$, we have $\psi(z, y^0) > \psi(y^0, y^0)$ but $\psi(z, x') =$ $-z < 1/2 - x' = \psi(x', x')$.

Thus, we cannot find any $y^0 \in [0, 1]$ and any neighborhood \mathcal{V}_0 of 0 such that $\psi(z, x') > \psi(x', x')$ for every deviation profile z that is upset by y^0 for all $x' \in \mathcal{V}_x$. Therefore, ψ is not recursively diagonal transfer continuous on X_0 , and thus, by Theorem 4.1, there is no symmetric pure strategy Nash equilibrium on X.

Besides, since all the games in Examples 3.1-3.5 are quasisymmetric, it is even easier to show the existence/nonexistence of pure strategy (symmetric) Nash equilibrium by working on a single payoff function ψ , instead of the aggregate payoff function U that is the sum of individual payoff functions.

Similar to Corollary 3.1, the following corollary is reached.

COROLLARY 4.1 Suppose a game $G = (X_i, u_i)_{i \in I}$ is quasi-symmetric, compact, and deviational transitive. Then it possesses a pure strategy symmetric Nash equilibrium if and only if $\psi(x, y)$ defined by (4) is 1-recursively diagonal transfer continuous on X.

We now provide some sufficient conditions for a game to be deviational transitive and 1recursively diagonal transfer continuous.

DEFINITION 4.2 A game $G = (X_i, u_i)_{i \in I}$ is diagonally monotonic if for each $\bar{x} \in X_0$, $u_i(\bar{x}, \ldots, x, \ldots, \bar{x})$ is either: (i) decreasing in x on $X_0 \setminus \bar{x}$ or (ii) increasing in x on $X_0 \setminus \bar{x}$. The following propositions provide sufficient conditions for a game $G = (X_i, u_i)_{i \in I}$ to be deviational transitive and 1-recursively diagonal transfer continuous.

PROPOSITION 4.1 Suppose X_i is a subset of \mathbb{R} . If a game $G = (X_i, u_i)_{i \in I}$ is diagonally monotonic, ψ is deviational transitive.

PROOF. We only need to show the case where ψ is nondecreasing in x. The proof of the case where ψ is non-increasing in x is similar.

We need to show that $\psi(z, y) > \psi(y, y)$ and $\psi(y, x) > \psi(x, x)$ imply that $\psi(z, x) > \psi(x, x)$. Indeed, when $\psi(y, x) > \psi(x, x)$, i.e., $u_i(x, \ldots, y, \ldots, x) > u_i(x, \ldots, x)$, we have y > x by monotonicity of $u_i(x, \ldots, y, \ldots, x)$. When $\psi(z, y) > \psi(y, y)$, we have z > y by monotonicity of $u_i(x, \ldots, y, \ldots, x)$. Thus we have z > y > x. Then, by monotonicity of $u_i(x, \ldots, y, \ldots, x)$, we have

$$u_i(x,\ldots,z,\ldots,x) > u_i(x,\ldots,y,\ldots,x) > u_i(x,\ldots,x)$$

and therefore $\psi(y, x) = u_i(x, ..., z, ..., x) > \psi(x, x)$, which means ψ is deviational transitive.

PROPOSITION 4.2 Suppose X_i is a subset of \mathbb{R} and a game $G = (X_i, u_i)_{i \in I}$ is diagonally monotonic. Any of the following conditions implies it is 1-recursively diagonal transfer continuous in x.

PROOF. We only need to prove the case of upper semi-continuity and the case where ψ is increasing in x. The proof of the case where ψ is decreasing in x is similar.

Suppose $\psi(y, x) > \psi(x, x)$ for $x, y \in X_0$. We need to show that there exists a point $y^0 \in X_0$ and a neighborhood \mathcal{V}_x of x such that $\psi(z, \mathcal{V}_x) > \psi(\mathcal{V}_x, \mathcal{V}_x)$ whenever $\psi(z, y^0) > \psi(y^0, y^0)$. Indeed, since $\psi(y, x) > \psi(x, x)$, we have y > x by diagonal monotonicity of ψ . Let $y^0 = x + \delta < y$ for some positive $\delta > 0$. We have $\psi(y^0, x) > \psi(x, x)$ by diagonal monotonicity of ψ . Then, by upper semi-continuity, there is a neighborhood $\mathcal{V}_x = \{x' \in X_0 : |x' - x| < \epsilon\}$ such that $\psi(y^0, x) > \psi(x', x)$ for all $x' \in \mathcal{V}_x$. Since $u_i(\bar{x}, \dots, x, \dots, \bar{x})$ is nondecreasing in x on $X_0 \setminus \bar{x}$ for all $\bar{x} \in X_0$, we particularly have $\psi(y^0, x') > \psi(x', x')$ for all $\bar{x} = x' \in \mathcal{V}_x$. Thus, whenever $\psi(z, y^0) > \psi(y^0, y^0)$, we have $z > y^0$ by diagonal monotonicity of ψ , and therefore, we have $\psi(z, x') > \psi(y^0, x') > \psi(x', x')$ for all $x' \in \mathcal{V}_x$, which means ψ is 1-recursively diagonal transfer continuous in x. **EXAMPLE** 4.3 (BAYE AND KOVENOCK; BAYE, TIAN, AND ZHOU) Consider the two-player quasi-symmetric game studied by Baye and Kovenock (1993), and Baye, Tian, and Zhou (1993). Two duopolists have zero costs and set prices (p_1, p_2) on $Z = [0, T] \times [0, T]$. The payoff functions are (for 0 < c < T):

$$u_i(p_1, p_2) = \begin{cases} p_i & \text{if } p_i \le p_{-i} \\ p_i - c & \text{otherwise} \end{cases}$$

One can interpret the game as a duopoly in which each firm has committed to pay brand loyal consumers a penalty of c if the other firm beats its price.⁶ These payoffs are neither quasiconcave nor continuous. However, the game is diagonally monotonic and upper semicontinuous, and thus it is 1-recursively diagonal transfer continuous. Thus, by Corollary 4.1, this game possesses a symmetric pure strategy equilibrium.

We end this section by discussing how Theorem 4.1 covers Reny's result as a corollary.

The game $G = (X_i, u_i)_{i \in I}$ is diagonally quasiconcave if X_i is convex, and for every player *i*, all $x^1, \ldots, x^m \in X$ and all $x \in co\{x^1, \ldots, x^m\}, u_i(x, \ldots, x) \ge \min_{1 \le k \le m} u_i(x, \ldots, x^k, \ldots, x).$

DEFINITION 4.3 A game $G = (X_i, u_i)_{i \in I}$ is diagonally better-reply secure if, whenever $(x^*, u^*) \in X \times \mathbb{R}$ is in the closure of the graph of its payoff function and (x^*, \ldots, x^*) is not an equilibrium, there is some player $i, y \in X_0$, and an open neighborhood \mathcal{V}_{x^*} of x^* such that $\psi(y, x') > \psi(x^*)$ for all $x' \in \mathcal{V}_{x^*}$.

COROLLARY 4.2 (RENY (1999)) If $G = (X_i, u_i)_{i \in I}$ is quasi-symmetric, compact, diangonally quasiconcave, and better-reply secure, then it possesses a symmetric pure strategy Nash equilibrium.

5 Full Characterization of Mixed Strategy Nash Equilibria

In this section, we fully characterize the existence of mixed strategy Nash equilibrium as corollaries to the pure strategy existence results derived in the previous sections. We assume throughout this section that each u_i is both bounded and measurable, and X_i is a compact Hausdorff space so we call $G = (X_i, u_i)_{i \in I}$ a compact, Hausdorff game. Consequently, if M_i denotes the set of (regular, countably additive) probability measures on the Borel subsets of X_i , M_i is compact in

⁶See Baye and Kovenock (1993) for an alternative formulation with both brand loyal and price conscious consumers, whereby a firm commits to pay a penalty if it does not provide the best price in the market.

the weak^{*} topology. Extend each u_i to $\prod_{i \in I} M_i$ by defining $u_i(\mu) = \int_X u_i(x) d\mu$ for all $\mu \in M$, and let $\overline{G} = (M_i, u_i)_{i \in I}$ denote the mixed extension of G, where $M = \prod_{i \in I} M_i$.

The definitions of recursive diagonal transfer continuity, etc. given in the previous sections, apply in the obvious ways to the mixed extension \bar{G} .

We now present the mixed strategy implications of Theorem 3.1.

THEOREM 5.1 Suppose that $G = (M_i, u_i)_{i \in I}$ is a compact, Hausdorff game. Then G possesses a mixed strategy Nash equilibrium if and only if its mixed extension, \overline{G} , is recursively diagonal transfer continuous.

This theorem strictly generalizes all the existence results on the mixed strategy equilibrium in the literature such as those in Nash (1950), Glicksberg (1952), Mas-Colell (1984), Dasgupta and Maskin (1986), Robson (1994), Simon (1987), Reny (1999), Monteiro and Page (2007), and Nessah and Tian (2008). Any sufficient conditions imposed in the existing theorems on the existence of mixed strategy Nash equilibrium imply the recursive diagonal transfer continuity of $\overline{U}(\cdot)$. To illustrate this, we present here the results of Monteiro and Page (2007), and Nessah and Tian (2008) as corollaries of Theorem 5.1. We first state some definitions introduced by them.

Monteiro and Page (2007) introduce the concept of uniform payoff security for games with compact separable metric strategy spaces and payoffs bounded and measurable in players' strategies. They show that if a game is compact and uniformly payoff secure, then its mixed extension \bar{G} is payoff secure.

DEFINITION 5.1 The game G is uniformly payoff secure if for every player $i \in I$, every $x_i \in X_i$, and every $\epsilon > 0$, there is a strategy $\overline{x}_i \in X_i$ such that for every $y_{-i} \in X_{-i}$, there exists a neighborhood $\mathcal{V}_{y_{-i}}$ of y_{-i} such that $u_i(\overline{x}_i, x'_{-i}) \ge u_i(x_i, y_{-i}) - \epsilon$, for all $x'_{-i} \in \mathcal{V}_{y_{-i}}$.

COROLLARY 5.1 [Monteiro and Page (2007)] If a game $G = (X_i, u_i)_{i \in I}$ is compact, bounded, separable metric, uniformly payoff secure, and has reciprocally upper semicontinuous payoffs, then it possesses a mixed strategy Nash equilibrium.

Nessah and Tian (2008) introduce the concept of uniform transfer continuity, and show that if a game $G = (X_i, u_i)_{i \in I}$ is uniformly transfer continuous, then the mixed extension \overline{G} is weakly transfer continuous.

DEFINITION 5.2 The game G is said to be *uniformly transfer continuous* if for every player $i \in I$, every $x_i \in X_i$, and every $\epsilon > 0$, there is a strategy $\overline{x}_i \in X_i$ such that for every $y_{-i} \in X_{-i}$, there exists a neighborhood $\mathcal{V}_{(x_i,y_{-i})}$ of (x_i, y_{-i}) such that

$$u_i(\overline{x}_i, x'_{-i}) + \epsilon \ge u_i(x_i, y_{-i}) \ge u_i(z) - \epsilon$$
, for all $z \in \mathcal{V}_{(x_i, y_{-i})}$.

COROLLARY 5.2 [Nessah and Tian (2008)] If a game $G = (X_i, u_i)_{i \in I}$ is compact, bounded, Hausdorff, and uniformly transfer continuous, then it possesses a mixed strategy Nash equilibrium.

We now provide a full characterization on the existence of symmetric mixed strategy Nash equilibrium for quasi-symmetric games. For the following result only, let M_0 denote the common set of mixed strategies for each player *i*.

Define an extended quasi-symmetric function $\bar{\psi}: M_0 \times M_0 \to \mathbb{R}$ by

$$\psi(\nu,\mu) = u_i(\mu,\dots,\nu,\dots,\mu). \tag{5}$$

Since \overline{G} is quasi-symmetric, μ^* is a symmetric mixed strategy Nash equilibrium if and only if $\overline{\psi}(\nu,\mu^*) \leq \overline{\psi}(\mu^*,\mu^*)$ for all $\nu \in M_0$.

THEOREM 5.2 Suppose that $G = (M_i, u_i)_{i \in I}$ is a compact, quasi-symmetric, and Hausdorff game. Then G possesses a symmetric mixed strategy Nash equilibrium if and only if its mixed extension payoff $\bar{\psi}(\nu, \mu)$ defined by (4) is recursively diagonal transfer continuous on M_0 .

This result covers Corollary 5.3 of Reny (1999) as a special case.

6 Full Characterization of Bayesian Nash Equilibrium

In this section, we provide a full characterization of Bayesian Nash equilibrium in an ex ante formulation of a Bayesian game, in which each player's beliefs are common prior. The existence of Bayesian Nash equilibrium in this formulation has been studied by Radner and Rosenthal (1982), Milgrom and Weber (1985), Vives (1990), and Zandt and Vives (2007). The full characterization of Bayesian Nash equilibrium in an interim or incomplete-information formulation of a Bayesian game studied by Van Zandt (2007) can be similarly investigated.

Let the strategy spaces be compact subsets of topological spaces and T_i the set of types of player *i*, a non-empty complete separable metric space. Denote by *T* the Cartesian product of the sets of types of the players, $T = \prod_{i \in I} T_i$. The common beliefs of the players are represented by μ , a probability measure on the Borel subsets of *T*. The measure μ_i will represent the marginal on *T*. The payoff to player *i* is given by $u_i : X \times T \to \mathbb{R}$, Borel measurable and bounded. A (pure) strategy for player *i* is a (Borel measurable) map $\alpha_i : T_i \to X_i$ that assigns an action to every possible type of the player. Let $\Sigma_i(\mu_i)$ denote the strategy space of player *i* when we identify strategies σ_i and u_i if they are equally μ_i -almost surely (a.s.)

Let

$$\Pi_i(\sigma) = \int_T u_i(\sigma_1(t_1), \dots, \sigma_n(t_n), t_i) d\mu(dt)$$
(6)

be the expected payoff to player i when agent j uses strategy $\sigma_j, j \in I$.

A strategy σ^* is a *Bayesian Nash equilibrium* of a game if

$$\Pi_i(\sigma^*) \ge \Pi_i(\sigma_i, \sigma^*_{-i}) \quad \forall \sigma_i \in \Sigma_i.$$

There are several results available in the literature on the existence of pure strategy equilibria in Bayesian games [e.g. Radner and Rosenthal (1982), Milgrom and Weber (1985), and Vives (1990)].

As a direct corollary of Theorem 3.1, the following result strictly generalizes all the existing results on the existence of Bayesian Nash equilibrium.

THEOREM 6.1 Suppose a Bayesian-Nash game $\Gamma = (\Sigma_i, \Pi_i)_{i \in I}$ is compact. Then it possesses a Bayesian Nash equilibrium if and only if Π_i is recursively diagonal transfer continuous on X.

This theorem strictly generalizes the existing results such as those in Ray and Rosenthal (1982), Milgrom and Weber (1985), Vives (1990), Athey (2001), Reny (2006), Van Zandt (2007), and Zandt and Vives (2008) as special cases.

7 Economic Applications

The approach and main result developed in the paper can also allow us to ascertain the existence of equilibria in important classes of economic games. As an application, in this section, we show how they can be employed to fully characterize the existence of competitive (or Walrasian) equilibrium for a certain class of economies.

One of the great achievements of economic theory in the last sixty years is the general equilibrium theory. The proof of the existence of a competitive equilibrium is generally considered one of the most important and robust results of economic theory. There are many different ways of establishing the existence of competitive equilibria, including the "excess demand approach" by showing that there is a price at which excess demand can be non-positive.

The significance of such an approach lies partly in the fact that demand and/or supply may not be continuous or even not be necessarily derived from profit maximizing behavior of price taking firms, but is determined by prices in completely different ways. It is well known that Walrasian equilibrium precludes the existence of an equilibrium in the presence of increasing returns to scale and assumes price-taking and profit-maximizing behavior. Some other alternative pricing rules then have been proposed such as loss-free, average cost, marginal cost, voluntary trading, and quantity-taking pricing rules in the presence of increasing returns to scale or more general types of non-convexities— cf. Beato (1982), Bonnisseau and Cornet (1990), Quinzii(1992), Tian

(2009) and the references therein. There is a large literature on the existence results using the excess demand approach, such as those in Gale (1955), Nikaido (1956, 1968, 1970), Debreu (1970, 1974, 1982), Sonnenschein (1972, 1973), Hildenbrand (1974), Hildenbrand and Kirman (1975), Grandmont (1977), Neuefeind (1980), Aliprantis and Brown (1983), Hüsseinov (1999), Momi(2003), Quah (2008), etc.

We provide a complete solution to the existence of competitive equilibrium in economies with general excess demand functions,⁷ in which commodities may be indivisible and excess demand functions may be discontinuous or do not have any structure except Walras' law. We introduce a condition, called *recursive transfer lower semi-continuity*, which is necessary and sufficient for the existence of general equilibrium in such economies. Thus, our result strictly generalizes all the existing results on the existence of equilibrium in economies with excess demand functions.

Let Δ be the closed L-1 dimensional unit simplex defined by

$$\Delta = \{ p \in \Re^L_+ : \sum_{l=1}^L p^l = 1 \},$$
(7)

and let $\hat{z}(\cdot) : \Delta \to \mathbb{R}^L \cup \{\pm \infty\}$ denote the excess demand function of some economy. A very important property of excess demand function is Walras' law, which can take either of the following two forms. The strong form of Walras' law is given by

$$p \cdot \hat{z}(p) = 0$$
 for all $p \in \Delta$,

and the weak form of Walras' law is given by

$$p \cdot \hat{z}(p) \leq 0$$
 for all $p \in \Delta$.

A price vector p^* is a *competitive or Walrasian* equilibrium if $\hat{z}(p^*) \leq 0$.

The equilibrium price problem is to find a price vector p which clears the markets for all commodities (i.e., the excess demand functions $\hat{z}(p) \leq 0$ for the free disposal equilibrium price or $\hat{z}(p) = 0$) under the assumption of Walras' law.

We say that price p upsets price q if p gives a higher value to q's excess demand, i.e. $p \cdot \hat{z}(q) > q \cdot \hat{z}(q) > 0$.

DEFINITION 7.1 (Recursive Upset Pricing) Let $\hat{z}(\cdot) : \Delta \to \mathbb{R}^L \cup \{\pm \infty\}$ be an excess demand function. We say that a non-equilibrium price vector $p^0 \in \Delta$ is *recursively upset by* $p \in \Delta$ if there exists a finite set of price vectors $\{p^1, p^2, \ldots, p\}$ such that $p^1 \cdot \hat{z}(p^0) > 0, p^2 \cdot \hat{z}(p^1) > 0, \ldots, p \cdot \hat{z}(p^{m-1}) > 0.$

⁷In the case of strictly convex preferences and production sets, we obtain excess demand functions rather than correspondences.

In words, a non-equilibrium price vector p^0 is recursively upset by p means that there exist finite upsetting price vectors p^1, p^2, \ldots, p^m with $p^m = p$ such that p^0 's excess demand is not affordable at p^1 , p^1 's excess demand is not affordable at p^2 , and p^{m-1} 's excess demand is not affordable at p^m . When the strong form of Walras' law holds, this implies that p^0 is upset by p^1 , p^1 is upset by p^2, \ldots, p^{m-1} is upset by p.

DEFINITION 7.2 (Recursive Transfer Lower Semi-Continuity) An excess demand function $\hat{z}(\cdot)$: $\Delta \to \mathbb{R}^L \cup \{\pm \infty\}$ is said to be *recursively transfer lower semi-continuous* on Δ if, whenever $q \in \Delta$ is not a competitive equilibrium price vector, there exists some price $p^0 \in \Delta$ (possibly $p^0 = q$) and a neighborhood \mathcal{V}_q such that $p \cdot \hat{z}(\mathcal{V}_q) > 0$ for any p that recursively upsets p^0 .

Roughly speaking, recursive transfer lower semi-continuity of $\hat{z}(\cdot)$ means that, whenever q is not a competitive equilibrium price vector, there exists another non-competitive equilibrium price vector p^0 such that all excess demands in some neighborhood of q are not affordable at any price vector p that recursively upsets p^0 . This implies that, if a competitive equilibrium fails to exist, then there is some non-equilibrium price vector q such that for every other price vector p^0 and every neighborhood of q, excess demand of some price vector q' in the neighborhood becomes affordable at price vector p that recursively upsets p^0 .

REMARK 7.1 While continuity does not imply nor is implied by recursive diagonal transfer continuity, recursive transfer lower semi-continuity is weaker than lower semi-continuity. Indeed, when $\hat{z}(\cdot)$ is lower semi-continuous, $p \cdot \hat{z}(\cdot)$ is also lower semi-continuous for any nonnegative vector p, and thus we have $p^m \cdot \hat{z}(q') > 0$ for all $q' \in N(q)$ and $p \in \Delta$.

Now we have the following theorem that strictly generalizes all the existing results on the existence of competitive equilibrium in economies that have single-valued excess demand functions.

THEOREM 7.1 Suppose an excess demand function $\hat{z}(\cdot) : \Delta \to \mathbb{R}^L \cup \{\pm \infty\}$ satisfies either of the two forms of Walras' law. Then there exists a competitive price equilibrium $p^* \in \Delta$ if and only if $\hat{z}(\cdot)$ is recursively transfer lower semi-continuous on Δ .

PROOF. Sufficiency (\Leftarrow). Define a function $\phi : \Delta \times \Delta \to \Re$ by $\phi(p,q) = p \cdot \hat{z}(p)$ for $p, q \in \Delta$. Since $p \cdot \hat{z}(q) > 0$ and $q \cdot \hat{z}(q) \le 0$ for all $p \in \Delta$ by Walras' law, we have $\phi(p,q) > \phi(q,q)$ for all $p, q \in \Delta$. Then, by recursive transfer lower semi-continuity of $\hat{z}(\cdot)$, ϕ is recursively diagonal transfer continuous on Δ .⁸ Thus, by the sufficiency of Theorem 3.1, there exists $p^* \in \Delta$ such

⁸The reverse may not be true under the weak form of Walras' law. However, when the strong form of Walras' law holds, \hat{z} is recursively transfer lower semi-continuous if and only if ϕ is recursively diagonal transfer continuous.

that $p \cdot \hat{z}(p^*) = \phi(p, p^*) \le \phi(p^*, p^*) \le 0$ for all $p \in \Delta$. Letting $p^1 = (1, 0, ..., 0), p^2 = (0, 1, 0, ..., 0)$, and $p^L = (0, 0, ..., 0, 1)$, we have $\hat{z}^l(p^*) \le 0$ for l = 1, ..., L and thus p^* is a competitive price equilibrium.

Necessity (\Rightarrow). Suppose p^* is a competitive price equilibrium and $p \cdot \hat{z}(q) > 0$ for $q, p \in \Delta$. Let $p^0 = p^*$ and N(q) be a neighborhood of q. Since $p \cdot \hat{z}(p^*) \leq 0$ for all $p \in \Delta$, it is impossible to find any sequence of finite price vectors $\{p^1, p^2, \ldots, p^m\}$ such that $p^1 \cdot \hat{z}(p^0) > 0, p^2 \cdot \hat{z}(p^1) > 0, \ldots, p^m \cdot \hat{z}(p^{m-1}) > 0$. Hence, the recursive transfer lower semi-continuity holds trivially.

8 Conclusion

The existing results only give sufficient conditions for the existence of equilibrium, and no complete solution to the question of the existence of equilibrium in general games has been given in the literature. This paper fills this gap by providing a full characterization of equilibrium in games with arbitrary strategy spaces and payoffs. We fully characterize the existence of pure strategy Nash equilibrium in games with general topological strategy spaces that may be discrete, continuum or non-convex and payoff functions that may be discontinuous or do not have any form of quasi-concavity. We establish a condition, called *recursive diagonal transfer continuity*, which is both necessary and sufficient for the existence of pure strategy Nash equilibrium in games with arbitrary compact strategy spaces and payoffs. As such, it strictly generalizes all the existing theorems on the existence of pure strategy Nash equilibrium. Recursive diagonal transfer continuity also permits full characterization results on the existence of symmetric pure strategy, mixed strategy Nash, and Bayesian Nash equilibria in games with general strategy spaces and payoffs. As an application of our approach and main result, we also fully characterizes the existence of competitive equilibrium for economies with excess demand functions.

We end the paper by remarking that characterization results are mainly for the purpose of identifying whether or not a game has an equilibrium, but not whether it is easy to check. Recursive diagonal transfer continuity provides a way of understanding equilibrium, more than necessarily providing a way to check its existence. Even so, in the paper, we use many known economic examples to illustrate that it is useful to employ recursive diagonal transfer continuity to check the existence of equilibrium, especially the nonexistence of pure strategy Nash equilibrium. Nevertheless, Nessah and Tian (2008) develop some very weak sufficient conditions that are relatively easy to check and generalize most of the existing results for the existence of equilibrium in discontinuous games. A potential future work may be attempted to find ways of applying recursive diagonal transfer continuity as a useful tool for establishing equilibrium.

References

- Aliprantis, C. D. and Brown, J. (1983): Equilibria in Markets with A Riesz Space of Commodities. *Journal of Mathematical Economics* 11, 189-207.
- Athey, S. (2001): Single Crossing Properties and the Existence of Pure Strategy Equilibria in Games of Incomplete Information, *Econometrica*, **69**, 861–889.
- Bagh, A., Jofre, A. (2006): Reciprocal Upper Semicontinuous and Better Reply Secure Games: A Comment. *Econometrica*, 74, 1715–1721.
- Barelli, P. and Soza, I. (2009): On the Existence of Nash Equilibria in Discontinuous and Qualitative Games, mimeo.
- Baye, M.R., Tian, G., Zhou, J. (1993): Characterizations of the Existence of Equilibria in Games with Discontinuous and Non-Quasiconcave Payoffs. *The Review of Economic Studies*, **60**, 935– 948.
- Beato, P. (1982): The Existence of Marginal Cost Pricing Equilibrium with Increasing Returns, *Quarterly Journal of Economics* 389, 669-688.
- Bonnisseau, J. M., Cornet, B. (1990): Existence of Marginal Cost Pricing Equilibrium in An Economy with Several Nonconvex Firms, *Econometrica* 58, 661-682.
- Carmona, G. (2005): On the Existence of Equilibria in Discontinuous Games: Three Counterexamples. *International Journal of Game Theory*, **33**, 181–187.
- Dasgupta, P., Maskin, E. (1986): The Existence of Equilibrium in Discontinuous Economic Games, I: Theory. *The Review of Economic Studies*, **53**, 1–26.
- Dutta B, and Sen, A. (1991): A Necessary and Sufficient Condition for Two-Person Nash Implementation, *The Review of Economic Studies*, 58, 121–128.
- Debreu, G. (1952): A Social Equilibrium Existence Theorem. *Proceedings of the National* Academy of Sciences of the U. S. A., 38.
- Debreu, G. (1974): Economies with a Finite Set of Equilibria, Econometrica, 38, 387-392.
- Debreu, G. (1974): Excess Demand Functions, Journal of Mathematical Economics 1, 15-21.
- Debreu, G. (1982): "Existence of Competitive Equilibrium" in Arrow, K. and M. Intrilligator (eds.), *Handbook of Mathematical Economics* (vol II). (Amsterdam: North Holland Publishing Company).

- Fudenberg, D., and Levine, D. K. (1983): Subgame-Perfect Equilibria of Finite and Infinite Horizon Games, *Journal of Economic Theory*, 31, 251-268.
- Gale, D. (1955): The Law of Supply and Demand, Mathematica Scandinavica 3, 155-169.
- Glicksberg, I.L. (1952): A Further Generalization of the Kakutani Fixed Point Theorem. *Proceed*ings of the American Mathematical Society, **3**, 170–174.
- Grandmont, J. M. (1977): Temporary General Equilibrium Theory, Econometrica 45, 535-572.
- Hildenbrand, W. (1974): *Core and Equilibria of a Large Economy*. (Princeton: Princeton University Press).
- Hildenbrand, W. and A.P. Kirman (1988): Equilibrium Analysis. (Amsterdam: North Holland).
- Hüsseinov, F. (1999): Boundary Behavior of Excess Demand and Existence of Equilibrium, *Journal of Economic Theory* 87, 434–449.
- Jackson, M. O. (2005): Non-existence of Equilibrium in Vickrey, Second-price, and English Auctions, working paper, Stanford University.
- Jones, A. J. (1980): *Game Theory: Mathematical Models of Conflict*. (Chichester: Ellis Harward Limited).
- Karlin, S. (1959): Mathematical Methods and Theory in Games, Programming and Economics, Vol. II (London: Pregamon Press).
- Maskin, E. (1999): Nash Equilibrium and Welfare Optimality, *Review of Economic Studies* 66, 23-38.
- Milgrom, P., and Roberts, H. (1990): Rationalizability, Learning, and Equilibrium in Games with Strategic Complementarities, *Econometrica* 58, 1255-1277.
- Milgrom, P., Weber, R. (1985): Distributional Strategies for Games with Incomplete Information, Mathematics of Operations Research 10, 619-632.
- Momi, T. Excess demand Functions with Incomplete Markets^aA Global Result, *Journal of Economic Theory* 111, 240[°]C250.
- Monteiro, P. K., Page, F. H. Jr. (2007): Uniform Payoff Security and Nash Equilibrium in Compact Games. *Journal of Economic Theory*, **134**, 566–575.

- Moore J., and Repullo, R. (1990): Nash Implementation: A Full Characterization, *Econometrica* 58, 1083–1099.
- Morgan, J., Scalzo, V. (2007): Pseudocontinuous Functions and Existence of Nash Equilibria. Journal of Mathematical Economics, 43, 174–183.
- Nash, J. (1950): Equilibrium Points in n-Person Games. Proceedings of the National Academy of Sciences, 36, 48–49.
- Nash, J. (1951): Noncooperative Games. Annals of Maths, 54, 286-295.
- Nessah, R., and Tian, G. (2008): The Existence of Equilibria in Discontinuous and Nonconvex Games, working paper.
- Nishimura, K., Friedman, J. (1981): Existence of Nash Equilibrium in *n*-Person Games without Quasi-Concavity. *International Economic Review*, **22**, 637–648.
- Negishi, T. (1962): The Stability of a Competitive Equilibrium: A Survey Article, *Econometrica* 30, 635-669.
- Neuefeind, W. (1980): Notes on Existence of Equilibrium Proofs and the Boundary Behavior of Supply, *Econometrica*, 1831-1837.
- Nikaido, H., and Isoda, K. (1955): Note on Noncooperative Convex Games, *Pacific J. Math.* 5, 807-815.
- Nikaido, H. (1956): On the Classical Multilateral Exchange Problem, *Metroeconomica* 8, 135-145.
- Nikaido, H. (1968): Convex Structures and Economic Theory. (New York: Academic Press).
- Nikaido, H. (1970): *Introduction to Sets and Mappings in Modern Economics*. (Amsterdam: North-Holland Publishing Company).
- Owen, G. (1968): Game Theory. (Philadelphia: W.B. Saunders Company).
- Quan, J. K.-H. (2008): The Existence of Equilibrium when Excess Demand Obeys the Weak Axiom, *Journal of Mathematical Economics* 44, 337[°]C343.
- Quinzii, M. (1992): Increasing Returns and Efficiency, (New York: Oxford University Press).
- Radner, R. and Rosenthal, R. (1982): Private Information and Pure-strategy Equilibria, Mathematics of Operations Research 7, 401-409.

Rahman, D. (2008): The Alternative to Equilibrium Existence, working paper.

- Reny, J. P. (1999): On the Existence of Pure and Mixed Strategy Nash Equilibria in Discontinuous Games, *Econometrica*, **67**, 1029–1056.
- Reny, P. J. (2006): On the Existence of Monotone Pure Strategy Equilibria in Bayesian Games, University of Chicago.
- Roberts, J. and Sonnenschein, H. (1977): "On the Existence of Cournot Equilibrium Without Concave Profit Functions," *Econometrica*, 45, 101 113.
- Robson, A. J. (1994): An Informationally Robust Equilibrium in Two-Person Nonzero-Sum Games, *Games and Economic Behavior*, 2, 233-245.
- Rosen, J. B. (1965): Existence and Uniqueness of Equilibrium Point for Concave n-Person Games, *Econometrica*, 33, 520–534.
- Simon, L. (1987): Games with Discontinuous Payoffs, Review of Economic Studies, 54, 569-597.
- Simon, L., Zame, W. (1990): Discontinuous Games and Endogenous Sharing Rules, *Econometrica*, **58**, 861–872.
- Sonnenschein, H. (1972): Market Excess Demand Functions, Econometrica 40, 549-563.
- Sonnenschein, H. (1973): Do Walras' Identity and Continuity Characterize the Class of Community Excess Demand Functions? *Journal of Economic Theory* 6, 345–354.
- Tarski, A. (1955): A Lattice-Theoretical Fixpoint Theorem and Its Applications, *Pacific J. Math.* 5, 285-309.
- Tian, G. (1992): Generalizations of the KKM Theorem and Ky Fan Minimax Inequality, with Applications to Maximal Elements, Price Equilibrium, and Complementarity, *Journal of Mathematical Analysis and Applications*, 170, 457–471.
- Tian, G. (1993): Necessary and Sufficient Conditions for Maximization of a Class of Preference Relations, *Review of Economic Studies*, **60**, 949–958.
- Tian, G. (2009): Implementation in Economies with Non-Convex Production Technologies Unknown to the Designer, *Games and Economic Behavior* 66, 526-545.
- Tian, G. and Zhou, J. (1992): The Maximum Theorem and the Existence of Nash Equilibrium of (Generalized) Games Without Lower Semicontinuities, *Journal of Mathematical Analysis and Applications*, 166, 351-364.

- Tian, G. and Zhou, Z. (1995): Transfer Continuities, Generalizations of the Weierstrass Theorem and Maximum Theorem: A Full Characterization. *Journal of Mathematical Economics*, 24, 281–303.
- Topkis, D. M.(1979): Equilibrium Points in Nonzero-Sum n-Person Submodular Games. *SIAM Journal on Control and Optimization* 17(6), 773–787.
- Varian, H. (1980): A Model of Sales, American Economic Review, 651-659.
- Van Zandt, T. (2007): Interim Bayesian Nash Equilibrium on Universal Type Spaces for Supermodular Games, Working Paper.
- Van Zandt, T. and Vives, X. (2007): Monotone Equilibria in Bayesian Games of Strategic Complementarities, *Journal of Economic Theory* 134, 339-360.
- Vives, X. (1990): Nash Equilibrium with Strategic Complementarities, *Journal of Mathematical Economics*, **19**, 305–321.
- Zhou, J. Zhou, X. and Tian, G. (1992): Transfer Method for Characterizing the Existence of Maximal Elements of Binary Relations on Compact or Noncompact Sets, SIAM Journal on Optimization, 2, 360–375.